UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, DC 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported) March 8, 2024

KORU Medical Systems, Inc. (Exact name of registrant as specified in its charter)

<u>Delaware</u>	<u>0-12305</u>	<u>13-3044880</u>
(State or other jurisdiction	(Commission	(IRS Employer
of incorporation)	File Number)	Identification No.)
	100 Corporate Drive, Mahwah, NJ	07430
	(Address of principal executive offices)	(Zip Code)
Reg	gistrant's telephone number, including area code	<u>(845) 469-2042</u>
	(Former name or former address, if changed since	e last report)
	The Form 8-K filing is intended to simultaneous (see General Instruction A.2. below):	usly satisfy the filing obligation of the registrant
[_] Written communications pursua	nt to Rule 425 under the Securities Act (17 CFR 2	230.425)
Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)		
Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))		
Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))		
Securities registered pursuant to Sec	tion 12(b) of the Act:	
<u>Title of each class</u> common stock, \$0.01 par val		ne of each exchange on which registered The Nasdaq Stock Market
Indicate by check mark whether the registrant is an emerging growth company as defined in as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).		
		Emerging growth company [_]
If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.		

ITEM 1.01 ENTRY INTO A MATERIAL DEFINITIVE AGREEMENT.

On March 8, 2024, KORU Medical Systems, Inc. (the "Company") entered into a loan and security agreement (the "Loan and Security Agreement"), by and between the Company and HSBC Ventures USA Inc., as lender providing for a revolving credit facility in an aggregate principal amount not to exceed \$5,000,000 (the "Revolver") and a term loan facility in an aggregate principal amount not to exceed \$5,000,000 (the "Term Loan" and collectively with the Revolver, the "Credit Facility"). The Company has not drawn on the Credit Facility, and there is no obligation for the Company to do so at any time.

Borrowings under the Credit Facility are secured by a first-priority lien on substantially all of the assets of the Company, subject to customary exceptions. The Revolver matures on December 31, 2025 and the Term Loan matures on December 1, 2028.

Borrowings under the Revolver will bear interest at the greater of Prime or 6.50%, payable in arrears on a monthly basis and at maturity. Borrowings under the Term Loan will bear interest at the greater of Prime minus 0.50% or 6.50% and will be interest-only through December 31, 2025, followed by 24 equal monthly payments of principal plus interest.

The Company will pay a final payment fee of 3.25% of the amount advanced under the Term Loan upon termination or maturity. The Company will pay a commitment fee of 0.25% of the total Revolver upon closing of the Credit Facility and an unused line fee of 0.20% per annum based on the daily unused amount of commitments under the Revolver, payable quarterly in arrears. The Company may use the proceeds of borrowings under the Credit Facility to pay transaction fees and expenses, provide for its working capital needs and for other general corporate purposes.

At the Company's option, the Company may at any time prepay the outstanding principal balance of the Term Loan in whole but not in part, with a prepayment fee of 2% of the advanced amount if prepaid in the first year of the Term Loan, 1% if prepaid in the second year of the Term Loan and 0% thereafter (which prepayment fee will be waived if the Term Loan is refinanced by the lender).

The Loan and Security Agreement contains customary affirmative covenants for transactions of this type, including, among others, the provision of financial and other information to the lender, notice to the lender upon the occurrence of certain material events, preservation of existence, maintenance of properties and insurance, compliance with laws, including environmental laws, the provision of additional guarantees, and an affiliate transactions covenant, subject to customary exceptions. The Loan and Security Agreement contains customary negative covenants, including, among others, restrictions on the ability to merge and consolidate with other companies, incur indebtedness, grant liens or security interests on assets, make investments, acquisitions, loans, or advances, pay dividends, and sell or otherwise transfer assets, all as subject to customary exceptions.

The Loan and Security Agreement contains a financial maintenance covenant that requires the Company to maintain a minimum Adjusted Quick Ratio (defined as the ratio of the Company's (i) unrestricted and unencumbered cash and cash equivalents maintained with the lender and its affiliates, plus eligible accounts receivable, to (ii) current liabilities) of not less than 1.50 to 1.00 tested on the last day of each month. The Loan and Security Agreement also provides for a number of customary events of default, including, among others: payment defaults to the lenders; voluntary and involuntary bankruptcy proceedings; covenant defaults; material inaccuracies of representations and warranties; certain change of control events; material money judgments; and other customary events of default. The occurrence of an event of default could result in the acceleration of obligations and the termination of lending commitments under the Loan and Security Agreement.

As a condition to the closing of the Term Loan, the Company issued to HSBC Ventures USA Inc. a Stock Purchase Warrant to purchase up to 76,104 shares of the Company's common stock, par value \$.01 per share, at an exercise price of \$2.12 per share. The warrant is exercisable for a period of ten years from issuance and vests as follows: 50% upon issuance and 50% upon any advance pursuant to the Term Loan.

The foregoing descriptions of the Loan and Security Agreement and Stock Purchase Warrant are qualified in their entirety by reference to the full text of the Loan and Security Agreement and Stock Purchase Warrant attached as Exhibits 10.1 and 10.2 to this Current Report on Form 8-K, respectively, which are incorporated by reference into this Item 1.01.

ITEM 2.02 RESULTS OF OPERATIONS AND FINANCIAL CONDITION.

On March 13, 2024, KORU Medical Systems, Inc. ("KORU Medical" or "we") issued a press release announcing its financial results for the fourth quarter and year ended December 31, 2023 and financial guidance for the fiscal year ended December 31, 2024. A related conference call will be held on March 13, 2023 at 4:30 pm Eastern Time.

KORU Medical is making reference to non-GAAP financial measures in both the press release and the conference call. Our management believes that investors' understanding of KORU Medical's performance is enhanced by disclosing the non-GAAP financial measures of Adjusted EBITDA, Adjusted Diluted EPS and Adjusted Gross Margin (each as defined below) as a reasonable basis for comparison of our ongoing results of operations. KORU Medical strongly encourages investors to review its consolidated financial statements and publicly filed reports in their entirety and cautions investors that the non-GAAP measures used by KORU Medical may differ from similar measures used by other companies, even when similar terms are used to identify such measures. Non-GAAP measures should not be considered replacements for, and should be read together with, the most comparable GAAP financial measures. A reconciliation of GAAP to non-GAAP results is provided in the attached Exhibit 99.1 press release.

We define Adjusted EBITDA as earnings (net (loss)/income) before interest expense/(income), net, income tax (benefit)/expense (including valuation allowance for deferred tax assets), depreciation and amortization, reorganization charges, discontinued product expense, manufacturing initiative expenses and stock-based compensation expense. We believe that Adjusted EBITDA is used by investors and other users of our financial statements as a supplemental financial measure that, when viewed with our GAAP results and the accompanying reconciliation, we believe provides additional information that is useful to gain an understanding of the factors and trends affecting our business. We also believe the disclosure of Adjusted EBITDA helps investors meaningfully evaluate and compare our cash flow generating capacity from quarter to quarter and year to year. Adjusted EBITDA is used by management as a supplemental internal measure for planning and forecasting overall expectations and for evaluating actual results against such expectations.

We present diluted earnings per share after eliminating items that we believe are not part of our ordinary operations and affect the comparability of the periods presented ("Adjusted Diluted EPS"). Adjusted Diluted EPS includes adjustments for depreciation and amortization, reorganization charges, discontinued product expense, manufacturing initiative expenses and stock-based compensation expense. We believe adjustments for these items allow investors to better understand our underlying operating results and facilitate comparisons between the periods shown. Management uses Adjusted Diluted EPS as a supplemental internal measure for planning and forecasting overall expectations and for evaluating actual results against such expectations.

Adjusted Gross Margin is defined as Net Loss excluding valuation allowance for deferred tax assets, stated as a percentage of net revenues. Management believes trends in the underlying profitability of the Company's business can be more clearly identified without the impact of the valuation allowance.

The information in Item 2.02 of this Current Report on Form 8-K, including Exhibit 99.1, is furnished and shall not be deemed "filed" for the purposes of Section 18 of the Securities Exchange Act of 1934, as amended.

ITEM 2.03 CREATION OF A DIRECT FINANCIAL OBLIGATION OR AN OBLIGATION UNDER AN OFF-BALANCE SHEET ARRANGEMENT OF A REGISTRANT.

The information set forth in Item 1.01 of this Current Report on Form 8-K is incorporated by reference into this Item 2.03 insofar as it relates to the creation of a direct financial obligation of the Company.

ITEM 3.02 UNREGISTERED SALES OF EQUITY SECURITIES.

The disclosure set forth in Item 1.01 of this Current Report on Form 8-K is incorporated by reference into this Item 3.02 insofar as it relates to the Stock Purchase Warrant.

ITEM 9.01 FINANCIAL STATEMENTS AND EXHIBITS.

Exhibits.

Exhibit No.	<u>Description</u>
10.1+	Loan and Security Agreement dated as of March 8, 2024 by and between KORU Medical Systems, Inc. and HSBC Ventures USA Inc. +
10.2	Stock Purchase Warrant issued to HSBC Ventures USA Inc. issued on March 8, 2024
99.1	Press Release dated March 13, 2024
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

⁺ Schedules and exhibits have been omitted pursuant to Item 601(b)(10) of Regulation S-K. The Registrant agrees to furnish supplementally a copy of any omitted schedule or exhibit to the U.S. Securities and Exchange Commission upon request.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

KORU Medical Systems, Inc.

(Registrant)

Date: March 13, 2024

By: /s/ Linda Tharby
Linda Tharby

President and Chief Executive Officer

- 5 -

LOAN AND SECURITY AGREEMENT

dated as of March 8, 2024

between

KORU MEDICAL SYSTEMS, INC.,

as Borrower

and

HSBC VENTURES USA INC.,

as Bank

This **LOAN AND SECURITY AGREEMENT** (this "Agreement") is dated as of March 8, 2024 (the "Effective Date") by and between **KORU MEDICAL SYSTEMS, INC.**, a Delaware corporation ("Borrower") and **HSBC VENTURES USA INC.** ("Bank"), and provides the terms on which Bank shall lend to Borrower and Borrower shall repay Bank. The parties agree as follows:

Section 1 **Definitions**.

- 1.1 <u>Definitions.</u> As used in this Agreement, the following capitalized terms have the following meanings:
- "Account" is any "account" as defined in the Code with such additions to such term as may hereafter be made, and includes, without limitation, all accounts receivable and other sums owing to any Loan Party.
- "Account Debtor" is any "account debtor" as defined in the Code with such additions to such term as may hereafter be made.
- "Acquisition" is any transaction, or any series of related transactions, consummated on or after the Effective Date, by which any Loan Party or a Subsidiary (a) acquires any going business or all or substantially all of the assets of any Person, whether through purchase of assets, merger or otherwise or (b) directly or indirectly acquires (in one transaction or as the most recent transaction in a series of transactions) at least a majority (in number of votes) of the equity interests of a Person which has ordinary voting power for the election of directors or other similar management personnel of a Person (other than equity interests having such power only by reason of the happening of a contingency) or a majority of the outstanding equity interests of a Person.
- "Adjusted Quick Ratio" is, calculated on a consolidated basis with respect to Borrower and its Subsidiaries, the ratio of (a) Quick Assets to (b) Current Liabilities.
- "Advance" or "Advances" is or are a revolving credit loan (or revolving credit loans) under the Revolving Line.
- "Affected Financial Institution" is (a) any EEA Financial Institution or (b) any UK Financial Institution.
- "Affiliate" is, with respect to any Person, each other Person that owns or controls directly or indirectly the Person, any Person that controls or is controlled by or is under common control with the Person. For purposes of this Agreement, in all cases HSBC Bank USA, N.A. shall be deemed to be an Affiliate of Bank.

- "Agreement" is defined in the preamble hereof.
- "Applicable Law" means all applicable provisions of constitutions, laws, statutes, ordinances, rules, treaties, regulations, permits, licenses, approvals, interpretations and orders of courts or Governmental Authorities and all orders and decrees of all courts and arbitrators.
- "Audited Financial Statements" are, as of any date, the most recent audited financial statements submitted to Bank pursuant to Section 6.2(f).
- "Authorized Signer" is any individual listed in any Loan Party's Borrowing Resolution who is authorized to execute the Loan Documents, including any Credit Extension request, on behalf of such Loan Party.
- "Availability Amount" is the lesser of (a) the Revolving Line or (b) the Borrowing Base, minus the sum of all outstanding principal amounts of any Advances.

- 1 -

- "Bank" is defined in the preamble hereof.
- "Bank Entities" is defined in Section 12.9.
- "Bank Expenses" are all (a) reasonable and documented out-of-pocket expenses incurred by Bank (including the reasonable and documented fees, charges and disbursements of counsel for Bank), and all fees and time charges and disbursements for attorneys who may be employees of Bank, in connection with the preparation, negotiation, execution, delivery, and administration of this Agreement and the other Loan Documents, or any amendment, modification or waiver of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated), (b) reasonable audit fees and documented related costs and expenses, and (c) out-of-pocket expenses incurred by Bank (including the fees, charges and disbursements of any counsel for Bank) and all fees and time charges for attorneys who may be employees of Bank, in connection with the enforcement or protection of its rights (i) in connection with this Agreement and the other Loan Documents (including without limitation, in connection with the enforcement of any judgment and in connection with any appeals), or (ii) in connection with the Credit Extensions made, including all such out-of-pocket expenses incurred, whether before or after a Default or Event of Default has occurred under any of the Loan Documents, relating to a workout, restructuring or other negotiations with any Loan Party respect of such Credit Extensions.
- "Bank Services" are any Letters of Credit, deposit accounts, Cash Management Services, and foreign exchange services (including any FX Contracts) previously, now, or hereafter provided to any Loan Party or any of its respective Subsidiaries by Bank and/or any Affiliate of Bank, as any such products or services may be identified in any of Bank's and/or Bank's Affiliates' various agreements and/or arrangements related thereto (each, a "Bank Services Agreement").
- "Beneficial Ownership Certification" is a certification regarding beneficial ownership required by the Beneficial Ownership Regulation.
 - "Beneficial Ownership Regulation" is 31 C.F.R. § 1010.230.
 - "Board" is Borrower's board of directors or equivalent governing body.
- "Books" are all of the Loan Parties' books and records including ledgers, federal and state tax returns, records regarding such Loan Party's assets or liabilities, the Collateral, business operations or financial condition, and all computer programs or storage or any equipment containing such information.
 - "Borrower" is defined in the preamble hereof.
- "Borrowing Base" is eighty percent (80.0%) of Eligible Accounts, as determined by Bank from Borrower's most recent Borrowing Base Certificate (and as may subsequently be updated by Bank based upon information received by Bank including, without limitation, Accounts that are paid and/or billed following the date of the Borrowing Base Certificate); provided, however, that Bank has the right to decrease the foregoing percentage in its commercially reasonable discretion to mitigate the impact of events, conditions, contingencies, or risks which may adversely affect the Collateral or its value.
 - "Borrowing Base Certificate" is that certain certificate in the form attached hereto as Exhibit C.
- "Borrowing Resolutions" are, with respect to any Person, those resolutions adopted by such Person's board of directors or equivalent governing body (and, if required under the terms of such Person's Operating Documents, equityholders) and delivered by such Person to Bank approving the Loan Documents to which such Person is a party and the transactions contemplated thereby, together with a certificate executed by its secretary on behalf of such Person certifying (a) such Person has the authority to execute, deliver, and perform its obligations under each of the Loan Documents to which it is a party, (b) that set forth as a part of or attached as an exhibit to such certificate is a true, correct, and complete

copy of the resolutions then in full force and effect authorizing and ratifying the execution, delivery, and performance by such Person of the Loan Documents to which it is a party, (c) the name(s) of the Person(s) authorized to execute the Loan Documents, including any Credit Extension request, on behalf of such Person, together with a sample of the true signature(s) of such Person(s), and (d) that Bank may conclusively rely on such certificate unless and until such Person shall have delivered to Bank a further certificate canceling or amending such prior certificate.

"Business Day" is any day that is not a Saturday, Sunday or a day on which Bank is closed, and if any determination of a "Business Day" shall relate to an FX Contract, the term "Business Day" shall mean a day on which dealings are carried on in the country of settlement of the Foreign Currency. In addition, when used in connection with a Credit Extension, the term "Business Day" is any such day that is also a day on which dealings in Dollar deposits are conducted by and between banks in the London interbank market.

"Cash Equivalents" are (a) marketable direct obligations issued or unconditionally guaranteed by the United States or any agency or any State thereof having maturities of not more than one (1) year from the date of acquisition; (b) commercial paper maturing no more than one (1) year after its creation and rated at least A-2 by Standard & Poor's Financial Services LLC, a subsidiary of S&P Global Inc., or at least P-2 by Moody's Investors Service, Inc., and any of their respective successors; (c) any certificates of deposit (or time deposit represented by a certificate of deposit), overnight bank deposit or banker's acceptance maturing no more than one (1) year after such time, or any overnight Federal funds transaction; and (d) money market accounts or mutual funds at least ninety-five percent (95.0%) of the assets of which constitute Cash Equivalents of the types described in clauses (a) through (c) of this definition.

"Cash Management Services" means any agreement, arrangement, or facility under which Bank or any Affiliate of Bank provides any of the following products or services to any of the Loan Parties:
(a) cash management services for collections, treasury management services (including controlled disbursement, overdraft, automated clearing house fund transfer services, return items and interstate depository network services), any demand deposit, payroll, trust or operating account relationships, credit cards (including, without limitation, commercial credit cards, virtual cards, purchasing cards and business debit cards), non-card e-payables services, and other cash management services, including electronic funds transfer services, lockbox services, stop payment services and wire transfer services.

"Change in Control" means (a) at any time, any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act) shall become, or obtain rights (whether by means of warrants, options or otherwise) to become, the "beneficial owner" (as defined in Rules 13(d)-3 and 13(d)-5 under the Exchange Act), directly or indirectly, of thirty percent (30.0%) or more of the ordinary voting power for the election of directors, partners, managers and members, as applicable, of Borrower (determined on a fully diluted basis) other than by the sale of Borrower's equity securities in a public offering or to venture capital or private equity investors so long as Borrower identifies to Bank the venture capital or private equity investors at least seven (7) Business Days prior to the closing of the transaction and provides to Bank a description of the material terms of the transaction; (b) during any period of twelve (12) consecutive months, a majority of the members of the Board cease to be composed of individuals (i) who were members of that board or equivalent governing body on the first (1st) day of such period, (ii) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in clause (i) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body or (iii) whose election or nomination to that board or other equivalent governing body was approved by individuals referred to in clauses (i) and (ii) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body; or (c) at any time, Borrower shall cease to own and control, of record and beneficially, directly or indirectly, one hundred percent (100.0%) of each class of outstanding stock, partnership,

- 3 -

membership, or other ownership interest or other equity securities of each Subsidiary of Borrower free and clear of all Liens (except Permitted Liens).

"Change in Law" is the occurrence, after the date of this Agreement, of any of the following:
(a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation, implementation or application thereof by any Governmental Authority or (c) the making or issuance of any request, rule, guideline or directive (whether or not having the force of law) by any Governmental Authority; provided that notwithstanding anything herein to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (ii) all requests, rules, guidelines or directives promulgated by Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Law", regardless of the date enacted, adopted or issued.

"Code" is the Uniform Commercial Code, as the same may, from time to time, be enacted and in effect in the State of New York; provided, that, to the extent that the Code is used to define any term herein or in any Loan Document and such term is defined differently in different Articles or Divisions of the Code, the definition of such term contained in Article or Division 9 shall govern; provided, further, that in the event that, by reason of mandatory provisions of law, any or all of the attachment, perfection, or priority of, or remedies with respect to, Bank's Lien on any Collateral is governed by the Uniform Commercial Code in effect in a jurisdiction other than the State of New York, the term "Code" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes of the provisions thereof relating to such attachment, perfection, priority, or remedies and for purposes of definitions relating to such provisions.

"Collateral" is any and all properties, rights and assets of the Loan Parties described on $\underline{\text{Exhibit}}$ \underline{A} .

"Collateral Account" is any Deposit Account, Securities Account, or Commodity Account.

"Collections Account" is defined in Section 6.8(a).

"Commodity Account" is any "commodity account" as defined in the Code with such additions to such term as may hereafter be made.

"Compliance Certificate" is that certain certificate in the form attached hereto as Exhibit B.

"Contracts" are software licensing subscription contracts (but not service contracts) executed with customers in the ordinary course of the Loan Parties' business.

"Control Agreement" is any control agreement entered into among the depository institution at which a Loan Party maintains a Deposit Account or the securities intermediary or commodity intermediary at which a Loan Party maintains a Securities Account or a Commodity Account, such Loan Party, and Bank pursuant to which Bank obtains control (within the meaning of the Code) over such Deposit Account, Securities Account, or Commodity Account.

"Copyrights" are any and all copyright rights, copyright applications, copyright registrations and like protections in each work of authorship and derivative work thereof, whether published or unpublished and whether or not the same also constitutes a trade secret.

"Credit Extension" is any Advance, Overadvance, Term Loan Advance, or any other extension of credit by Bank and/or an Affiliate of Bank for Borrower's benefit.

- 4 -

"Current Liabilities" are (a) all obligations and liabilities of Borrower and its Subsidiaries to Bank, plus (b) without duplication, the aggregate amount of Borrower and its Subsidiaries' Total Liabilities that mature within one (1) year.

"Default" is any event which with notice or passage of time or both, would constitute an Event of Default.

"Default Rate" is defined in Section 2.4(b).

"Deferred Revenue" is all amounts received or invoiced in advance of performance under contracts and not yet recognized as revenue, pursuant to GAAP.

"Deposit Account" is any "deposit account" as defined in the Code with such additions to such term as may hereafter be made.

"Designated Deposit Account" is the account denominated in Dollars, account number *******, maintained by Borrower with Bank's Affiliate.

"Division" means, in reference to any Person which is an entity, the division of such Person into two (2) or more separate Persons, with the dividing Person either continuing or terminating its existence as part of such division, including, without limitation, as contemplated under Section 18-217 of the Delaware Limited Liability Company Act for limited liability companies formed under Delaware law, Section 17-220 of the Delaware Revised Uniform Limited Partnership Act for limited partnerships formed under Delaware law, or any analogous action taken pursuant to any other Applicable Law with respect to any corporation, limited liability company, partnership or other entity.

"Dollars," "dollars" or use of the sign "\$" are only lawful money of the United States and not any other currency, regardless of whether that currency uses the "\$" sign to denote its currency or may be readily converted into lawful money of the United States.

"Dollar Equivalent" is, at any time, (a) with respect to any amount denominated in Dollars, such amount, and (b) with respect to any amount denominated in a Foreign Currency, the equivalent amount therefor in Dollars as determined by Bank at such time on the basis of the then-prevailing rate of exchange in New York, New York, for sales of the Foreign Currency for transfer to the country issuing such Foreign Currency.

"Domestic Subsidiary" is a Subsidiary organized under the laws of the United States or any state or territory thereof or the District of Columbia other than (a) any Subsidiary all or substantially all of whose assets consist of equity securities of (or debt obligations owed or treated as owed by such) controlled foreign corporations (as defined in Section 957 of the IRC) and (b) any Subsidiary that is a direct or indirect Subsidiary of a Subsidiary that is a controlled foreign corporation (as defined in Section 957 of the IRC).

"**Draw Period**" is the period of time commencing upon the Effective Date and continuing through the earlier to occur of (a) March 31, 2025 or (b) an Event of Default.

"EEA Financial Institution" is (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a Subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" is any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

- 5 -

"EEA Resolution Authority" is any public administrative authority or any Person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution.

"Effective Date" is defined in the preamble hereof.

"Eligible Accounts" means Accounts owing to Borrower which arise in the ordinary course of Borrower's business that meet all Borrower's representations and warranties in Section 5.3, that have been, at the option of Bank, confirmed in accordance with Section 5.3 of this Agreement, and are due and owing from Account Debtors deemed creditworthy by Bank in its sole discretion. Bank reserves the right, at any time after the Effective Date, in its sole discretion in each instance, to adjust any of the criteria set forth below and to establish new criteria. Unless Bank otherwise agrees in writing, Eligible Accounts shall not include:

- (a) Accounts (i) for which the Account Debtor is Borrower's Affiliate, officer, employee, investor, or agent, or (ii) that are intercompany Accounts;
- (b) Accounts that the Account Debtor has not paid within ninety (90) days of invoice date regardless of invoice payment period terms;
- (c) Accounts with credit balances over ninety (90) days from invoice date, to the extent of such credit balances;
- (d) Accounts owing from an Account Debtor if fifty percent (50.0%) or more of the Accounts owing from such Account Debtor have not been paid within ninety (90) days of invoice date:
- (e) Accounts owing from an Account Debtor (i) which does not have its principal place of business in the United States or (ii) whose billing address (as set forth in the applicable invoice for such Account) is not in the United States, unless in the case of both (i) and (ii) such Accounts are otherwise approved by Bank in writing;
- (f) Accounts billed from and/or payable to Borrower outside of the United States (sometimes called foreign invoiced accounts);
- (g) Accounts in which Bank does not have a first priority, perfected security interest under all Applicable Law;
 - (h) Accounts billed and/or payable in a Currency other than Dollars;
- (i) Accounts owing from an Account Debtor to the extent that Borrower is indebted or obligated in any manner to the Account Debtor (as creditor, lessor, supplier or otherwise sometimes called "contra" accounts, accounts payable, customer deposits or credit accounts), but only to the extent of such Indebtedness or obligations;
- (j) Accounts with or in respect of accruals for marketing allowances, incentive rebates, price protection, cooperative advertising and other similar marketing credits, unless otherwise approved by Bank in writing, but only to the extent of such credits;
- (k) Accounts owing from an Account Debtor which is a United States government entity or any department, agency, or instrumentality thereof unless Borrower has assigned its

payment rights to Bank and the assignment has been acknowledged under the Federal Assignment of Claims Act of 1940, as amended;

- (l) Accounts owing from an Account Debtor which is a government entity or any department, agency, or instrumentality thereof;
- (m) Accounts with customer deposits and/or with respect to which Borrower has received an upfront payment, to the extent of such customer deposit and/or upfront payment;
- (n) Accounts for demonstration or promotional equipment, or in which goods are consigned, or sold on a "sale guaranteed", "sale or return", "sale on approval", or other terms if Account Debtor's payment may be conditional;
- (o) Accounts owing from an Account Debtor where goods or services have not yet been rendered to the Account Debtor (sometimes called memo billings or pre-billings);
- (p) Accounts subject to contractual arrangements between Borrower and an Account Debtor where payments shall be scheduled or due according to completion or fulfillment requirements (sometimes called contracts accounts receivable, progress billings, milestone billings, or fulfillment contracts);
- (q) Accounts owing from an Account Debtor the amount of which may be subject to withholding based on the Account Debtor's satisfaction of Borrower's complete performance (but only to the extent of the amount withheld; sometimes called retainage billings);
- (r) Accounts subject to trust provisions, subrogation rights of a bonding company, or a statutory trust;
- (s) Accounts owing from an Account Debtor that has been invoiced for goods that have not been shipped to the Account Debtor unless Bank, Borrower, and the Account Debtor have entered into an agreement acceptable to Bank wherein the Account Debtor acknowledges that (i) it has title to and has ownership of the goods wherever located, (ii) a bona fide sale of the goods has occurred, and (iii) it owes payment for such goods in accordance with invoices from Borrower (sometimes called "bill and hold" accounts);
 - (t) Accounts for which the Account Debtor has not been invoiced;
- (u) Accounts that represent non-trade receivables or that are derived by means other than in the ordinary course of Borrower's business;
- (v) Accounts for which Borrower has permitted Account Debtor's payment to extend beyond ninety (90) days (including Accounts with a due date that is more than ninety (90) days from invoice date);
- (w) Accounts arising from chargebacks, debit memos or other payment deductions taken by an Account Debtor;
- (x) Accounts arising from product returns and/or exchanges (sometimes called "warranty" or "RMA" accounts);

- 7 -

- (y) Accounts in which the Account Debtor disputes liability or makes any claim (but only up to the disputed or claimed amount), or if the Account Debtor is subject to an Insolvency Proceeding (whether voluntary or involuntary), or becomes insolvent, or goes out of business; and
- (z) Accounts owing from an Account Debtor with respect to which Borrower has received Deferred Revenue (but only to the extent of such Deferred Revenue);
- (aa) Accounts owing from an Account Debtor, whose total obligations to Borrower exceed twenty-five percent (25.0%) of all Accounts (other than Accounts for which the Account Debtor is McKesson, for which such percentage shall be fifty percent (50.0%)), for the amounts that exceed that percentage, unless Bank approves in writing; and
- (bb) Accounts for which Bank in its commercially reasonable discretion determines collection to be doubtful, including, without limitation, accounts represented by "refreshed" or "recycled" invoices.

"Equipment" is all "equipment" as defined in the Code with such additions to such term as may hereafter be made, and includes without limitation all machinery, fixtures, goods, vehicles (including motor vehicles and trailers), and any interest in any of the foregoing.

"ERISA" is the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" is any trade or business (whether or not incorporated) under common control with any Loan Party or Subsidiary thereof within the meaning of Section 414(b) or (c) of the Internal Revenue Code of 1986, as amended, (and Sections 414(m) and (o) thereof for purposes of provisions relating to Section 412 thereof).

"Event of Default" is defined in Section 8.

"Exchange Act" is the Securities Exchange Act of 1934, as amended.

"Excluded Accounts" is defined in Section 6.8(b).

"Excluded Taxes" means any of the following Taxes imposed on or with respect to Bank or required to be withheld or deducted from a payment to Bank: (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes and branch profits Taxes, in each case (i) imposed by the United States or as a result of Bank being organized under the laws of, or having its principal office or its applicable lending office located in, the jurisdiction imposing such tax (or any political subdivision thereof), or (ii) that are Other Connection Taxes, (b) U.S. federal withholding Taxes that are imposed on amounts payable to Bank to the extent that the obligation to withhold amounts existed on the date of this Agreement, and (c) any withholding Taxes imposed under FATCA.

"FATCA" means Sections 1471 through 1474 of the IRC, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not more onerous to comply with), any regulations or official interpretations thereof and any agreements entered into pursuant to Section 1471(b)(1) of the IRC, and any intergovernmental agreement, or provision of any treaty or convention entered into in connection with the implementation of such Sections of the Code and any fiscal or regulatory legislation, rules or practices adopted pursuant to such intergovernmental agreement, or provision of any treaty or convention.

"FCPA" is defined in Section 5.14.

- 8 -

"Final Payment" is a payment (in addition to and not in substitution for the regular monthly payments of principal plus accrued interest) equal to the aggregate original principal amount of each Term Loan Advance extended by Bank to Borrower hereunder, multiplied by three and one-quarter of one percent (3.25%), which payment shall be fully earned as of the Funding Date of each such Term Loan Advance, but payment shall be deferred until the earliest to occur of (a) the Term Loan Maturity Date, (b) the payment in full of the Term Loan Advances, (c) as required by Section 2.5(d) or Section 2.5(e), or (d) the termination of this Agreement.

"Foreign Currency" is lawful money of a country other than the United States.

"Foreign Subsidiary" is any Subsidiary which is not a Domestic Subsidiary.

"Funding Date" is any date on which a Credit Extension is made to or for the account of Borrower which shall be a Business Day.

"FX Contract" is any foreign exchange contract by and between a Loan Party and Bank and/or any Affiliate of Bank under which such Loan Party commits to purchase from or sell to Bank and/or any Affiliate of Bank a specific amount of Foreign Currency on a specified date.

"GAAP" is generally accepted accounting principles set forth in the opinions and pronouncements of the Accounting Principles Board of the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or in such other statements by such other Person as may be approved by a significant segment of the accounting profession, which are applicable to the circumstances as of the date of determination.

"General Intangibles" is all "general intangibles" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation, all Intellectual Property, all claims, income and other tax refunds, security and other deposits, payment intangibles, contract rights, options to purchase or sell real or personal property, rights in all litigation presently or hereafter pending (whether in contract, tort or otherwise), insurance policies (including without limitation key man, property damage, and business interruption insurance), payments of insurance and rights to payment of any kind.

"Governmental Approval" is any consent, authorization, approval, order, license, franchise, permit, certificate, accreditation, registration, filing or notice, of, issued by, from or to, or other act by or in respect of, any Governmental Authority, including, without limitation, Healthcare Permits.

"Governmental Authority" is any nation or government, any state or other political subdivision thereof, any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative functions of or pertaining to government, any securities exchange and any self-regulatory organization.

"Guarantee Agreement" is each agreement entered into by a Guarantor providing for the guaranty of the Obligations, in each case, in form and substance satisfactory to Bank.

"Guaranter" is each Subsidiary and other Person who guarantees the Obligations pursuant to a Guarantee Agreement.

"Healthcare Laws" means all applicable laws relating to the operation or management of Borrower's business, proper billing and collection practices relating to the payment for healthcare services, insurance law (including law related to payment for "no-fault" claims) and workers compensation law as they relate to the provision of, and billing and payment for, healthcare services, patient healthcare, patient healthcare information, patient abuse, personnel, operating policies, fee splitting, including, without limitation, (a) all federal and state fraud and abuse laws, including, without limitation, the federal Anti-Kickback Statute (42 U.S.C. §1320a-7b(b)), the Stark Law (42 U.S.C.

-9-

§1395nn), the civil False Claims Act (31 U.S.C. §3729 et seq.), the administrative False Claims Law (42 U.S.C. § 1320a-7b(a)), the Anti-Inducement Law (42 U.S.C. § 1320a-7a(a)(5)), the exclusion laws (42 U.S.C. § 1320a-7); (b) the Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic and Clinical Health Act of 2009; (c) the Medicare Regulations and the Medicaid Program (Title XIX of the Social Security Act); (d) quality, safety and accreditation standards and requirements of all applicable state laws or regulatory bodies; (e) all laws, policies, procedures, requirements and regulations pursuant to which Healthcare Permits are issued; (f) any laws, regulations or administrative guidance with respect to fee splitting by healthcare professionals and the corporate practice of medicine in any jurisdiction in which any Borrower or any Guarantor operates; and (g) any and all comparable state or local laws and other applicable health care laws, regulations, manual provisions, policies and administrative guidance, each of (a) through (g) as may be amended from time to time and the regulations promulgated pursuant to each such law.

"Healthcare Permit" means, with respect to any Person, a permit issued or required under Healthcare Laws applicable to the business of Borrower or, or necessary in the possession, ownership, warehousing, marketing, promoting, sale, labeling, furnishing, distribution or delivery of goods or services under Healthcare Laws applicable to the business of Borrower.

"HIPAA" means, collectively, the Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic Clinical Health (HITECH) Act and the implementing regulations thereto.

"Indebtedness" is, as to any Person at a particular time, without duplication, all of the following, whether or not included as indebtedness or liabilities in accordance with GAAP:

- (a) all obligations of such Person for borrowed money and all obligations of such Person evidenced by bonds, debentures, notes, loan agreements or other similar instruments;
- (b) all direct or contingent obligations of such Person arising under letters of credit (including standby and commercial), bankers' acceptances, guaranties, surety bonds and similar instruments:
- (c) all obligations of such Person in respect of the deferred purchase price of property or services (other than trade accounts payable in the ordinary course of business and not past due for more than sixty (60) days after the date on which such trade account was created);
- (d) all obligations of such Person under conditional sale or other title retention agreements relating to property acquired by such Person, including, but not limited to, indebtedness secured by (or for which the holder of such indebtedness has an existing right, contingent or otherwise, to be secured by) a Lien on property owned, acquired or being purchased by such Person (including indebtedness arising under conditional sales or other title retention agreements), whether or not such indebtedness shall have been assumed by such Person or is limited in recourse;
 - (e) all capital lease obligations of such Person; and
 - (f) all guarantees of such Person in respect of any of the foregoing.

"Indemnified Person" is defined in Section 12.3.

"Indemnified Tax" is defined in Section 12.10.

"Initial Audit" is Bank's inspection of Borrower's Accounts, the Collateral, and Borrower's Books, with results satisfactory to Bank in its sole discretion.

"Insolvency Proceeding" is any proceeding by or against any Person under the United States Bankruptcy Code, or any other bankruptcy or insolvency law, including assignments for the benefit of creditors, compositions, extensions generally with its creditors, or proceedings seeking reorganization, arrangement, or other relief.

"Intellectual Property" is, with respect to any Person, all of such Person's right, title, and interest in and to the following:

- its Copyrights, Trademarks and Patents;
- any and all trade secrets and trade secret rights, including, without limitation, any rights to unpatented inventions, know-how and operating manuals;
 - any and all source code and domain names;
 - any and all design rights which may be available to such Person; (d)
- any and all claims for damages by way of past, present and future infringement of any of the foregoing, with the right, but not the obligation, to sue for and collect such damages for said use or infringement of the Intellectual Property rights identified above; and
- (f) all amendments, renewals and extensions of any of the Copyrights, Trademarks or Patents.

"Inventory" is all "inventory" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation all merchandise, raw materials, parts, supplies, packing and shipping materials, work in process and finished products, including without limitation such inventory as is temporarily out of any Loan Party's custody or possession or in transit and including any returned goods and any documents of title representing any of the above.

"Investment" is any beneficial ownership interest in any Person (including stock, partnership interest or other securities), and any loan, advance or capital contribution to any Person.

"IRC" means the U.S. Internal Revenue Code of 1986, as amended.

"Key Person" is each of Borrower's (i) Chief Executive Officer, who is Linda Tharby as of the Effective Date and (ii) Chief Financial Officer, who is Thomas Adams as of the Effective Date.

"Lien" is a claim, mortgage, deed of trust, levy, charge, pledge, security interest or other encumbrance of any kind, whether voluntarily incurred or arising by operation of law or otherwise against any property.

"Loan Documents" are, collectively, this Agreement and any schedules, exhibits, certificates, notices, and any other documents related to this Agreement, including each Guarantee Agreement, any subordination agreement, the Warrant, any Control Agreement, any Bank Services Agreement, any pledge agreement, notes or guaranties executed by any Loan Party or any Subsidiary of a Loan Party and any other present or future agreement by any Loan Party or any Subsidiary of a Loan Party with or for the benefit of Bank and/or any Affiliate of Bank in connection with this Agreement and/or Bank Services, all as amended, restated, supplemented, or otherwise modified from time to time.

"Loan Parties" are Borrower and the Guarantors.

- 11 -

"Material Adverse Change" is (a) a material impairment in the perfection or priority of Bank's Lien in the Collateral or in the value of such Collateral; (b) a material adverse change in the business, assets, operations, properties, or financial condition of any Loan Party or any of its Subsidiaries; or (c) a material adverse change in the validity or enforceability of any of the Loan Documents and the rights and remedies of Bank thereunder.

"Net Cash" means (a) Borrower and its Subsidiaries' unrestricted and unencumbered cash maintained with Bank or Bank's Affiliates, or with financial institutions outside of Bank or Bank's Affiliates in accounts which are subject to a Control Agreement in favor of Bank, minus (b) Total Outstandings.

"Obligations" are the Loan Parties' obligations to pay when due any debts, principal, interest, fees, the Final Payment, the Prepayment Premium, the Termination Fee, the Unused Revolving Line Facility Fee, Bank Expenses, and other amounts any Loan Party owes Bank or any Affiliate of Bank now or later, whether under this Agreement, the other Loan Documents (other than the Warrant), any Bank Services Agreement or otherwise, including, without limitation other Bank Services, if any, and including interest accruing after Insolvency Proceedings begin and debts, liabilities, or obligations of any Loan Party assigned to Bank, and to perform any Loan Party's duties under the Loan Documents (other than the Warrant).

- "Operating Documents" are, for any Person, such Person's formation documents, and, (a) if such Person is a corporation, its bylaws in current form, (b) if such Person is a limited liability company, its limited liability company agreement (or similar agreement), and (c) if such Person is a partnership, its partnership agreement (or similar agreement), each of the foregoing with all current amendments or modifications thereto.
- "Other Connection Taxes" means, with respect to Bank, Taxes imposed as a result of a present or former connection between Bank and the jurisdiction imposing such tax (other than connections arising from Bank having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, or engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Loan Document).
- "Other Taxes" means all present or future stamp, court or documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, any Loan Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment.
 - "Overadvance" is defined in Section 2.3.
 - "Participant Register" is defined in Section 12.2.
- "Patents" are all patents, patent applications and like protections including without limitation improvements, divisions, continuations, renewals, reissues, extensions and continuations-in-part of the same.
- "Patriot Act" is the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)).
 - "Payment Date" is the first (1st) calendar day of each month.
 - "Perfection Certificate" is defined in Section 5.1.
 - "Permitted Indebtedness" is:

- 12 -

- (a) the Loan Parties' Indebtedness to Bank and/or any Affiliate of Bank under this Agreement, the other Loan Documents and any Bank Services Agreement;
- (b) Indebtedness existing on the Effective Date and shown on the Perfection Certificate;
 - (c) Subordinated Debt;
- (d) unsecured Indebtedness to trade creditors incurred in the ordinary course of business and not overdue by more than sixty (60) days;
- (e) Indebtedness incurred as a result of endorsing negotiable instruments received in the ordinary course of business;
- (f) Indebtedness secured by Liens permitted under clause (c) of the definition of "Permitted Liens" hereunder;
- (g) Indebtedness consisting of the financing of insurance premiums in the ordinary course of business; provided that if the insurance premiums financed increased by more than fifteen percent (15.0%) compared to the previous twelve (12) month period, such financing arrangement shall be approved in writing by Bank; and
- (h) extensions, refinancings, modifications, amendments and restatements of any item of Permitted Indebtedness (a) through (f) above, provided that the principal amount thereof is not increased; it is not secured by a Lien on any additional assets, the obligors remain the same, and the terms thereof are not otherwise modified to impose more burdensome terms upon Borrower or the applicable Subsidiary, as the case may be.

"Permitted Investments" are:

(a) Investments (including, without limitation, Subsidiaries) existing on the Effective Date and shown on the Perfection Certificate;

- (b) (i) Investments consisting of Cash Equivalents and (ii) any Investments permitted by Borrower's investment policy, as amended from time to time, provided that such investment policy (and any such amendment thereto) has been approved in writing by Bank;
- (c) Investments consisting of the endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of the Loan Parties; and
- (d) Investments consisting of Deposit Accounts in which Bank has a perfected security interest to the extent required by Section 6.8.

"Permitted Liens" are:

- (a) Liens existing on the Effective Date and shown on the Perfection Certificate or arising under this Agreement and the other Loan Documents and any Bank Services Agreement;
- (b) Liens for taxes, fees, assessments or other government charges or levies, either (i) not due or thereafter payable without any interest or penalty or (ii) being contested in good faith

- 13 -

and for which such Loan Party maintains adequate reserves on the Books, provided that no notice of any such Lien has been filed or recorded under the IRC, and the Treasury Regulations adopted thereunder;

- (c) Liens securing capital leases and purchase money Liens on Equipment acquired or held by Borrower or a Subsidiary incurred for financing the acquisition of such Equipment securing no more than One Hundred Thousand Dollars (\$100,000.00) in the aggregate amount outstanding;
- (d) Liens of carriers, warehousemen, landlords, mechanics, suppliers, or other Persons that are possessory in nature arising in the ordinary course of business so long as (i) such Liens attach only to Inventory, and (ii) secure liabilities in the aggregate amount not to exceed One Hundred Thousand Dollars (\$100,000.00) which are not delinquent or remain payable without penalty or which are being contested in good faith and by appropriate proceedings which proceedings have the effect of preventing the forfeiture or sale of the property subject thereto;
- (e) Liens to secure payment of workers' compensation, employment insurance, old-age pensions, social security and other like obligations incurred in the ordinary course of business (other than Liens imposed pursuant to ERISA);
- (f) Liens incurred in the extension, renewal or refinancing of the Indebtedness secured by Liens described in clauses (a) and (c) above, but any extension, renewal or replacement Lien must be limited to the property encumbered by the existing Lien and the principal amount of the Indebtedness secured thereby may not increase;
- (g) leases or subleases of real property granted in the ordinary course of Borrower's business (or, if referring to another Person, in the ordinary course of such Person's business), and leases, subleases, non-exclusive licenses or sublicenses of personal property (other than Intellectual Property) granted in the ordinary course of Borrower's business (or, if referring to another Person, in the ordinary course of such Person's business), if the leases, subleases, licenses and sublicenses do not prohibit granting Bank a security interest therein and do not interfere with the ordinary conduct of the business;
- (h) (i) non-exclusive licenses of Intellectual Property and (ii) licenses of Intellectual Property that are not material to the business of Borrower or its Subsidiaries, in each case, granted to their customers or commercial partners in the ordinary course of business and which do not interfere with the ordinary conduct of Borrower's or its Subsidiaries' respective businesses:
- (i) Liens arising from attachments or judgments, orders, or decrees in circumstances not constituting an Event of Default under Section 8.4 or Section 8.7;
- (j) Liens arising out of conditional sale, title retention, consignment or similar arrangements for sale of goods entered into in the ordinary course of business;
- (k) servitudes, easements, rights of way, restrictions and other similar encumbrances on real property imposed by applicable laws and encumbrances consisting of zoning or building restrictions, easements, licenses, restrictions on the use of property or minor imperfections in title thereto which, in the aggregate, are not material, and which do not in any case materially detract from the value of the property subject thereto or interfere with the ordinary conduct of the business; and
- (l) bankers' liens, rights of setoff and similar Liens incurred on deposits arising in the ordinary course of business; provided that (i) Bank has a first priority perfected security

interest in such account to the extent required in this Agreement and (ii) such account is permitted to be maintained pursuant to Section 6.8 of this Agreement;

- "**Person**" is any individual, sole proprietorship, partnership, limited liability company, joint venture, company, trust, unincorporated organization, association, corporation, institution, public benefit corporation, firm, joint stock company, estate, entity or government agency.
- "Prepayment Premium" shall be an additional fee, payable to Bank, with respect to the Term Loan Advances, in an amount equal to:
- (a) for a prepayment of the Term Loan Advances made on or prior to the first (1st) anniversary of the Effective Date, two percent (2.0%) of the outstanding principal amount of the Term Loan Advances immediately prior to the date of such prepayment;
- (b) for a prepayment of the Term Loan Advances made after the first (1st) anniversary of the Effective Date, but on or prior to the second (2nd) anniversary of the Effective Date, one percent (1.0%) of the outstanding principal amount of the Term Loan Advances immediately prior to the date of such prepayment; and
- (c) for a prepayment of the Term Loan Advances made after the second (2nd) anniversary of the Effective Date, zero percent (0.0%) of the outstanding principal amount of the Term Loan Advances immediately prior to the date of such prepayment.

Notwithstanding the foregoing, provided no Event of Default has occurred and is continuing, the Prepayment Premium shall be waived by Bank, if Bank or an Affiliate of Bank closes on the refinance and re-documentation of the Term Loan Advances (in its sole and absolute discretion) prior to the Term Loan Maturity Date.

"Prime Rate" is the rate of interest per annum equal to the "prime rate" as displayed on the Bloomberg screen page that displays such rate; provided that if such rate of interest does not appear on a page of the Bloomberg screen, on the appropriate page of such other information service or such other source that publishes such rate as shall be selected by Bank, then the "Prime Rate" shall mean the rate of interest per annum announced by Bank as its prime rate in effect at its principal office. The Prime Rate is a reference rate and does not necessarily represent the lowest or best rate actually charged to any customer. Any change in the Prime Rate shall take effect at the opening of business on the day after the public announcement of such change. Notwithstanding any terms in this Agreement to the contrary, if at any time such rate of interest is less than zero percent (0.0%) per annum, such rate shall be deemed to be zero percent (0.0%) per annum for purposes of this Agreement.

"Protected Person" is defined in Section 12.12.

"Quick Assets" is, on any date, Borrower and its Subsidiaries' unrestricted and unencumbered cash and Cash Equivalents maintained with Bank or Bank's Affiliates, plus (ii) Eligible Accounts.

"Register" is defined in Section 12.2.

"Registered Organization" is any "registered organization" as defined in the Code with such additions to such term as may hereafter be made.

"Requirement of Law" is as to any Person, the Operating Documents of such Person, and any law (statutory or common), treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.

- 15 -

- "Responsible Officer" is any of the Chief Executive Officer, Chief Financial Officer, Chief Technology Officer, and Vice President of Finance.
- "Restricted License" is any material license or other similar agreement with respect to which Borrower or any of its Subsidiaries is the licensee that validly prohibits or otherwise restricts Borrower or such Subsidiary from granting a security interest in Borrower's or such Subsidiary's interest in such license or agreement or any property subject to such license or similar agreement.
- "Revolving Line" is an aggregate principal amount not to exceed Five Million Dollars (\$5,000,000.00) outstanding at any time.

"Revolving Line Maturity Date" is December 31, 2025.

"Sanctions" is defined in Section 5.13.

"SEC" is the Securities and Exchange Commission, any successor thereto, and any analogous Governmental Authority.

"Securities Account" is any "securities account" as defined in the Code with such additions to such term as may hereafter be made.

"Streamline Balance" is defined in the definition of Streamline Period.

"Streamline Period" is, on and after the Effective Date, provided no Event of Default has occurred and is continuing, the period (a) commencing on the first day of the month following the day that Borrower provides to Bank a written report that Borrower has, for each consecutive day in the immediately preceding month, Net Cash, as determined by Bank in its commercially reasonable discretion, in an amount at all times greater than Five Million Dollars (\$5,000,000.00) (the "Streamline Balance"); and (b) terminating on the earlier to occur of (i) the occurrence of an Event of Default, and (ii) the first day thereafter in which Borrower fails to maintain the Streamline Balance, as determined by Bank in its commercially reasonable discretion. Upon the termination of a Streamline Period, Borrower shall maintain the Streamline Balance each consecutive day for one (1) fiscal quarter as determined by Bank in its commercially reasonable discretion, prior to entering into a subsequent Streamline Period. Borrower shall give Bank prior written notice of Borrower's election to enter into any such Streamline Period, and each such Streamline Period shall commence on the first day of the monthly period following the date Bank determines, in its commercially reasonable discretion, that the Streamline Balance has been achieved.

"Subordinated Debt" is unsecured indebtedness incurred by any Loan Party subordinated to all of the Loan Parties' now or hereafter indebtedness to Bank (pursuant to a subordination, intercreditor, or other similar agreement in form and substance satisfactory to Bank entered into between Bank and the other creditor), on terms and in amounts reasonably acceptable to Bank.

"Subsidiary" is, as to any Person, a corporation, partnership, limited liability company or other entity of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one (1) or more intermediaries, or both, by such Person. Unless the context otherwise requires, each reference to a Subsidiary herein shall be a reference to a Subsidiary of Borrower.

"Taxes" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

- 16 -

"Termination Fee" is defined in Section 2.5(d).

"Term Loan Advance" and "Term Loan Advances" are each defined in Section 2.2(c).

"Term Loan Amortization Date" means January 1, 2026.

"Term Loan Maturity Date" is December 1, 2028.

"**Total Liabilities**" is on any day, obligations that should, under GAAP, be classified as liabilities on Borrower and its Subsidiaries' consolidated balance sheet, including all Indebtedness.

``Total Outstandings" is, as of any date of determination, the outstanding principal balance of all Advances.

"Trademarks" are any trademark and servicemark rights, whether registered or not, applications to register and registrations of the same and like protections, and the entire goodwill of the business of a Loan Party connected with and symbolized by such trademarks.

"Transaction Request Form" is that certain form attached hereto as Exhibit D.

"Transfer" is defined in Section 7.1.

"**Transition Period**" means the period of time commencing on the Effective Date and continuing through the earlier to occur of (i) April 7, 2024, or (ii) an Event of Default.

"UK Bribery Act" is defined in Section 5.14.

"UK Financial Institution" is any BRRD Undertaking (as such term is defined under the PRA Rulebook (as amended from time to time) promulgated by the United Kingdom Prudential Regulation Authority) or any persons falling within IFPRU 11.6 of the FCA Handbook (as amended from time to time) promulgated by the United Kingdom Financial Conduct Authority, which includes certain credit institutions and investment firms, and certain affiliates of such credit institutions or investment firms.

"Unused Revolving Line Facility Fee" is defined in Section 2.5(e).

"Warrant" is that certain stock purchase warrant dated as of the Effective Date between Borrower and Bank, as amended, modified, supplemented and/or restated from time to time.

Terms Generally. The definitions of terms herein shall apply equally to the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "include", "includes" and "including" shall be deemed to be followed by the phrase "without limitation". The word "law" shall be construed as referring to all statutes, rules, regulations, codes and other laws (including official rulings and interpretations thereunder having the force of law or with which affected Persons customarily comply) and all judgments, orders and decrees of all Governmental Authorities. The word "will" shall be construed to have the same meaning and effect as the word "shall". Unless the context requires otherwise (a) any definition of or reference to any agreement, instrument or other document herein shall be construed as referring to such agreement, instrument or other document as from time to time amended, restated, supplemented or otherwise modified (subject to any restrictions on such amendments, restatements, supplements or modifications set forth herein), (b) any definition of or reference to any statute, rule or regulation shall be construed as referring thereto as from time to time amended, supplemented or otherwise modified (including by succession of comparable successor laws), (c) any reference herein to any Person shall be construed to include such Person's successors and assigns (subject to any restrictions on assignments set forth herein) and, in the case of any Governmental Authority, any other Governmental Authority that shall have succeeded to any or all functions thereof, (d) the words "herein", "hereof" and "hereunder", and words of similar import, shall be construed to refer to this

- 17 -

Agreement in its entirety and not to any particular provision hereof, (e) all references herein to Articles, Sections, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Exhibits and Schedules to, this Agreement, (f) any reference in any definition to the phrase "at any time" or "for any period" shall refer to the same time or period for all calculations or determinations within such definition, and (g) the words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract rights.

- 1.3 Accounting and Other Terms. Accounting terms not defined in this Agreement shall be construed following GAAP, and calculations and determinations must be made following GAAP (except with respect to unaudited financial statements for the absence of footnotes and subject to normal year-end audit adjustments). All other terms contained in this Agreement, unless otherwise indicated, shall have the meaning provided by the Code to the extent such terms are defined therein.
- 1.4 <u>Divisions by Limited Liability Companies</u>. For all purposes hereunder and under the other Loan Documents, and without limiting any restrictions or prohibitions in this Agreement, if in connection with any Division: (a) any asset, right, obligation or liability of any Person becomes the asset, right, obligation or liability of a different Person, then it shall be deemed to have been transferred from the original Person to the subsequent Person, and (b) any new Person comes into existence, such new Person shall be deemed to have been organized by the holders of its equity interests at such time on the first date of its existence for purposes of Section 6.12.

Section 2 Loan and Terms of Payment.

2.1 <u>Promise to Pay.</u> Borrower hereby unconditionally promises to pay Bank the outstanding principal amount of all Credit Extensions and accrued and unpaid interest thereon as and when due in accordance with this Agreement and any other applicable Loan Document.

2.2 Credit Extensions.

(a) Revolving Line.

- (i) <u>Availability.</u> Subject to the terms and conditions of this Agreement, Bank shall make Advances from time to time to Borrower prior to the Revolving Line Maturity Date in an aggregate outstanding amount not to exceed the Availability Amount. Amounts borrowed under the Revolving Line may be repaid and, prior to the Revolving Line Maturity Date, reborrowed, subject to the applicable terms and conditions precedent herein.
- (ii) <u>Termination; Repayment</u>. The Revolving Line terminates on the Revolving Line Maturity Date, when the principal amount of all Advances, the unpaid interest thereon, and all other Obligations relating to the Revolving Line shall be immediately due and payable.

(b) <u>Term Loan Advances</u>.

(i) <u>Availability.</u> Subject to the terms and conditions of this Agreement, upon Borrower's request, during the Draw Period, Bank shall make term loan advances in an aggregate principal amount not to exceed Five Million Dollars (\$5,000,000.00) (each, a "Term Loan Advance" and, collectively, the "Term Loan Advances"). Borrower may not request more than two (2) Term Loan Advances hereunder. After repayment, no Term Loan Advance (or any portion thereof) may be reborrowed.

- (ii) <u>Interest Period.</u> Commencing on the first (1st) Payment Date of the month following the month in which the Funding Date of the applicable Term Loan Advance occurs, and continuing on each Payment Date thereafter, Borrower shall make monthly payments of interest on the principal amount of each Term Loan Advance at the rate set forth in Section 2.4(b).
- (iii) <u>Repayment.</u> Commencing on the Term Loan Amortization Date, and continuing on each Payment Date thereafter, Borrower shall repay the Term Loan Advances in (i) twenty-four (24) consecutive equal monthly installments of principal, plus (ii) monthly payments of accrued interest at the rate set forth in Section 2.4(b). All outstanding principal and accrued and unpaid interest with respect to the Term Loan Advances, and all other outstanding Obligations with respect to the Term Loan Advances, are due and payable in full on the Term Loan Maturity Date.
- (iv) <u>Mandatory Prepayment Upon an Acceleration</u>. If the Term Loan Advances are accelerated following the occurrence of an Event of Default, Borrower shall immediately pay to Bank an amount equal to the sum of: (i) all outstanding principal plus accrued and unpaid interest, plus (ii) the Prepayment Premium, as applicable, (iii) Final Payment, and (iv) all other sums, if any, that shall have become due and payable, including interest at the Default Rate with respect to any past due amounts.
- (v) <u>Permitted Prepayment of Term Loan Advances</u>. Borrower shall have the option to prepay all, but not less than all, of the Term Loan Advances advanced by Bank under this Agreement, provided Borrower (i) provides written notice to Bank of its election to prepay the Term Loan Advances at least thirty (30) days prior to such prepayment, and (ii) pays, on the date of such prepayment (A) all outstanding principal plus accrued and unpaid interest, (B) the Prepayment Premium, as applicable, (C) the Final Payment, and (D) all other sums, if any, that shall have become due and payable, including interest at the Default Rate with respect to any past due amounts.
- 2.3 Overadvances. If, at any time, the sum of the Total Outstandings exceeds the lesser of (i) the Revolving Line or (ii) the Borrowing Base, Borrower shall immediately pay to Bank in cash the amount of such excess (such excess, the "Overadvance"). Without limiting Borrower's obligation to repay Bank any Overadvance, Borrower shall pay Bank interest on the outstanding amount of any Overadvance, on demand, at a rate per annum equal to the rate that is otherwise applicable to Advances plus five percent (5.0%).

2.4 <u>Payment of Interest on the Credit Extensions.</u>

(a) <u>Interest Rate</u>:

- (i) <u>Advances</u>. Subject to Section 2.4(b), the principal amount outstanding under the Revolving Line shall accrue interest at a floating per annum rate equal to the greater of (A) the Prime Rate and (B) six and one-half of one percent (6.50%), which interest, in each case, shall be payable monthly in accordance with Section 2.4(d) below.
- (ii) <u>Term Loan Advances</u>. Subject to Section 2.4(b), the principal amount outstanding under each Term Loan Advance shall accrue interest at a floating per annum rate equal to the greater of (A) the Prime Rate minus one-half of one percent (0.50%) and (B) six and one-half of one percent (6.50%), which interest, in each case, shall be payable monthly in accordance with Section 2.4(d) below. Notwithstanding any terms in this Agreement to the contrary, if at any time the interest rate applicable to any Term Loan Advance is less than zero percent (0.0%), such interest rate shall be deemed to be zero percent (0.0%) for all purposes of this Agreement.

- 19 -

- (b) <u>Default Rate</u>. Immediately upon the occurrence and during the continuance of an Event of Default, all Obligations shall bear interest at a rate per annum which is five percent (5.0%) above the rate that is otherwise applicable thereto (the "Default Rate"). Fees and expenses which are required to be paid by Borrower pursuant to the Loan Documents (including, without limitation, Bank Expenses) but are not paid when due shall bear interest until paid at a rate equal to the highest rate applicable to the Obligations inclusive of the Default Rate. Payment or acceptance of the increased interest rate provided in this Section 2.4(b) is not a permitted alternative to timely payment and shall not constitute a waiver of any Event of Default or otherwise prejudice or limit any rights or remedies of Bank.
- (c) <u>Adjustment to Interest Rate</u>. Changes to the interest rate of any Credit Extension based on changes to the Prime Rate shall be effective on the effective date of any change to the Prime Rate and to the extent of any such change.
- (d) <u>Payment; Interest Computation</u>. Interest is payable monthly on the first (1st) day of each calendar month and shall be computed on the basis of a year of three hundred sixty-five

(365) days (or three hundred sixty-six (366) days in a leap year), and in each case shall be payable for the actual number of days elapsed (including the first (1st) day but excluding the last day). In computing interest, (i) all payments received after 12:00 p.m. Eastern time on any day shall be deemed received at the opening of business on the next Business Day, and (ii) the date of the making of any Credit Extension shall be included and the date of payment shall be excluded; provided, however, that if any Credit Extension is repaid on the same day on which it is made, such day shall be included in computing interest on such Credit Extension.

- 2.5 <u>Fees</u>. Borrower shall pay to Bank:
- (a) <u>Revolving Line Commitment Fee</u>. A fully earned, nonrefundable commitment fee of Twelve Thousand Five Hundred Dollars (\$12,500.00), on the Effective Date;
 - (b) <u>Prepayment Premium</u>. The Prepayment Premium, when due hereunder;
 - (c) <u>Final Payment</u>. The Final Payment, when due hereunder;
- (d) <u>Termination Fee.</u> Upon termination of this Agreement for any reason prior to the Revolving Line Maturity Date, in addition to the payment of any other amounts then-owing, a termination fee (the "Termination Fee") in an amount equal to Fifty Thousand Dollars (\$50,000.00); provided that no termination fee shall be charged if the credit facility hereunder is refinanced or redocumented with a new facility from Bank or any Affiliate of Bank;
- (e) <u>Unused Revolving Line Facility Fee</u>. Payable quarterly in arrears, on the first (1st) day of each calendar quarter following the Effective Date and prior to the Revolving Line Maturity Date, and on the Revolving Line Maturity Date, a fee (the "Unused Revolving Line Facility Fee") in an amount equal to two-tenths of one percent (0.20%) per annum of the average unused portion of the Revolving Line, as determined by Bank. The unused portion of the Revolving Line, for purposes of this calculation, shall be calculated on a calendar year basis and shall equal the difference between (i) the Revolving Line, and (ii) the average for the period of the daily closing balance of the Revolving Line outstanding; and
- (f) <u>Bank Expenses</u>. All Bank Expenses incurred through and after the Effective Date, when due (or, if no stated due date, upon demand by Bank).

- 20 -

Unless otherwise provided in this Agreement or in a separate writing by Bank, Borrower shall not be entitled to any credit, rebate, or repayment of any fees earned by Bank pursuant to this Agreement notwithstanding any termination of this Agreement or the suspension or termination of Bank's obligation to make loans and advances hereunder. Bank may, at its option, deduct amounts owing by Borrower under the clauses of this Section 2.5 pursuant to the terms of Section 2.6(c). Bank shall provide Borrower written notice of deductions made from the Designated Deposit Account pursuant to the terms of the clauses of this Section 2.5.

2.6 <u>Payments; Application of Payments; Debit of Accounts.</u>

- (a) All payments to be made by Borrower under any Loan Document shall be made in immediately available funds in Dollars, without setoff or counterclaim, before 2:00 p.m. Eastern time on the date when due. Payments of principal and/or interest received after 2:00 p.m. Eastern time are considered received at the opening of business on the next Business Day. When a payment is due on a day that is not a Business Day, the payment shall be due the next Business Day, and additional fees or interest, as applicable, shall continue to accrue until paid.
- (b) Bank shall apply payments received with respect to the Obligations, first, to pay any fees, indemnities, or expense reimbursements then due to Bank, second, to pay interest then due and payable on any Credit Extension, third, to prepay principal on the Credit Extensions and to pay any amounts owing in respect of Bank Services, and fourth, to the payment of any other Obligation due to Bank from Borrower. Borrower shall have no right to specify the order or the accounts to which Bank shall allocate or apply any payments required to be made by Borrower to Bank or otherwise received by Bank under this Agreement when any such allocation or application is not specified elsewhere in this Agreement.
- (c) Bank may, at its option, debit any of Borrower's deposit accounts, including the Designated Deposit Account, for principal and interest payments or any other amounts Borrower owes Bank when due.

Section 3 Conditions of Credit Extensions.

3.1 <u>Conditions Precedent to Initial Credit Extension</u>. Bank's obligation to make the initial Credit Extension under this Agreement is subject to the condition precedent that Bank shall have received, in form and substance satisfactory to Bank, the following:

- (a) duly executed signatures to the Loan Documents to be entered into on the Effective Date, including this Agreement, any Control Agreements, the Warrant and the Perfection Certificate;
- (b) copies of the Operating Documents (including formation documents certified by the Secretary of State (or equivalent agency) of each Loan Party's jurisdiction of organization on a date that is no earlier than thirty (30) days prior to the Effective Date) and long-form good standing certificates of each Loan Party certified by the Secretary of State (or equivalent agency) of such Loan Party's jurisdiction of organization or formation and each jurisdiction in which such Loan Party is qualified to conduct business except where the failure to do so could not reasonably be expected to have a material adverse effect on such Loan Party's business, each as of a date no earlier than thirty (30) days prior to the Effective Date;
- (c) duly executed Control Agreement with HSBC Bank USA, N.A. with respect to Borrower's account maintained at HSBC Bank USA, N.A. ending ****;

- 21 -

- (d) completed Borrowing Resolutions for each Loan Party;
- (e) certified copies, dated as of a recent date, of financing statement searches and intellectual property searches, as Bank may request, accompanied by written evidence (including any UCC termination statements) that the Liens indicated in any such financing statements and intellectual property searches either constitute Permitted Liens or have been or, in connection with the initial Credit Extension, will be terminated or released;
- (f) copies of proper financing statements, to be filed on the Effective Date under the Code of all jurisdictions that Bank may deem necessary or desirable in order to perfect the Liens created hereunder, covering the Collateral;
- (g) evidence satisfactory to Bank that the insurance policies and endorsements required by Section 6.7 hereof are in full force and effect, together with appropriate evidence showing lender loss payable and/or additional insured clauses or endorsements in favor of Bank;
 - (h) with respect to the initial Advance, the completion of the Initial Audit;
- (i) with respect to the initial Advance, a duly executed Control Agreement with HSBC Bank USA, N.A. with respect to the Collections Account;
- (j) a capitalization table of Borrower and copies of Borrower's equity documents;
- (k) copies of the (i) documentation and other information requested by Bank in connection with applicable "know your customer" and anti-money-laundering rules and regulations, including, without limitation, the Patriot Act, in each case at least five (5) days prior to the Effective Date and (ii) at least three (3) Business Days prior to the Effective Date, if Borrower qualifies as a "legal entity customer" under the Beneficial Ownership Regulation, a Beneficial Ownership Certification;
- (l) Bank shall have completed a due diligence investigation of Borrower and its Subsidiaries in scope, and with results, satisfactory to Bank, and shall have been given such access to the management, records, books of account, contracts and properties of Borrower and its Subsidiaries and shall have received such financial, business and other information regarding each of the foregoing Persons and businesses as Bank shall have requested; and
- (m) upon Bank confirming in writing to Borrower that all other conditions precedent set forth in Sections 3.1 and 3.2 have been satisfied, payment of the fees and Bank Expenses then due as specified in Section 2.5 hereof, including any fees pursuant to Section 2.5.
- 3.2 <u>Conditions Precedent to all Credit Extensions</u>. Bank's obligations to make each Credit Extension under this Agreement, including the initial Credit Extension under this Agreement, are subject to the following conditions precedent:
- (a) timely receipt of an executed Transaction Request Form and Borrowing Base Certificate (to the extent required pursuant to Section 6.2(a));
- (b) the representations and warranties in this Agreement shall be true, accurate, and complete in all material respects on the date of the Transaction Request Form and on the Funding Date of each Credit Extension; provided, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof;

and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects (or all respects, as applicable) as of such date, and no Default or Event of Default shall have occurred and be continuing or result from the Credit Extension. Borrower's acceptance of each Credit Extension is Borrower's representation and warranty on that date that the representations and warranties in this Agreement remain true, accurate, and complete in all material respects; provided, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects (or all respects, as applicable) as of such date; and

- (c) Bank determines that there has not been any material impairment in the general affairs, management, results of operation, financial condition or the prospect of repayment of the Obligations, or any material adverse deviation by Borrower from the most recent business plan of Borrower presented to and accepted by Bank.
- 3.3 <u>Covenant to Deliver.</u> Borrower agrees to deliver to Bank each item required to be delivered to Bank under Sections 3.1 and 3.2 of this Agreement as a condition precedent to any Credit Extension, as applicable. Borrower expressly agrees that a Credit Extension made prior to the receipt by Bank of any such item shall not constitute a waiver by Bank of Borrower's obligation to deliver such item, and the making of any Credit Extension in the absence of a required item shall be in Bank's sole discretion.
- 3.4 <u>Procedures for Borrowing.</u> Subject to the prior satisfaction of all other applicable conditions to the making of a Credit Extension set forth in this Agreement, to obtain a Credit Extension, Borrower shall notify Bank (which notice shall be irrevocable) by electronic mail by 3:00 p.m. Eastern time one (1) Business Day prior to the Funding Date of such Credit Extension. In connection with such notification, Borrower must promptly deliver to Bank by electronic mail a completed Transaction Request Form and Borrowing Base Certificate, each executed by an Authorized Signer, together with such other reports and information as Bank may request in its sole discretion. Bank shall credit proceeds of a Credit Extension to the Designated Deposit Account. Bank may make Credit Extensions under this Agreement based on instructions from an Authorized Signer (it being understood that any deemed Credit Extensions pursuant to the terms hereof for purposes of satisfying Obligations that have become due shall not require any such instructions).

Section 4 Creation of Security Interest.

4.1 Grant of Security Interest.

- (a) Each Loan Party hereby grants Bank, to secure the payment and performance in full of all of the Obligations, a continuing security interest in, and pledges to Bank, the Collateral, wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof.
- (b) Each Loan Party acknowledges that it previously has entered, and/or may in the future enter, into Bank Services Agreements with Bank or any Affiliate of Bank. Regardless of the terms of any Bank Services Agreement, such Loan Party agrees that any amounts such Loan Party owes Bank or any Affiliate of Bank thereunder shall be deemed to be Obligations hereunder and that it is the intent of such Loan Party, Bank and Bank's Affiliates to have all such Obligations secured by the first priority perfected security interest in the Collateral granted herein (subject only to Permitted Liens).

- 23 -

- (c) If this Agreement is terminated, Bank's Lien in the Collateral shall continue until the Obligations (other than inchoate indemnity obligations) are repaid in full in cash. Upon payment in full in cash of the Obligations (other than inchoate indemnity obligations) and at such time as Bank's obligation to make Credit Extensions has terminated, Bank shall, at the sole cost and expense of Borrower, release its Liens in the Collateral and all rights therein shall revert to the applicable Loan Party. In the event (i) all Obligations (other than inchoate indemnity obligations), except for Bank Services, are satisfied in full, and (ii) this Agreement is terminated, Bank shall terminate the security interest granted herein upon Borrower providing cash collateral acceptable to Bank or any Affiliate of Bank in its good faith business judgment for Bank Services, if any.
- 4.2 <u>Priority of Security Interest</u>. Each Loan Party represents, warrants, and covenants that the security interest granted herein is and shall at all times continue to be a first priority perfected security interest in the Collateral (subject only to Permitted Liens). If any Loan Party shall acquire a commercial tort claim in excess of One Hundred Thousand Dollars (\$100,000.00) individually or One Hundred Fifty Thousand Dollars (\$150,000.00) in the aggregate, Borrower shall notify Bank of the general details thereof in the Compliance Certificate required to be delivered under this Agreement following such event and shall grant to Bank in a writing signed by the applicable Loan Party and delivered with such Compliance Certificate a security interest therein and in the proceeds thereof, all upon the terms of this Agreement, with such writing to be in form and substance reasonably satisfactory to Bank.

- 4.3 <u>Authorization to File Financing Statements</u>. Each Loan Party hereby authorizes Bank to file financing statements, without notice to such Loan Party, with all appropriate jurisdictions to perfect or protect Bank's interest or rights hereunder, including a notice that any disposition of the Collateral in violation of this Agreement, by any Loan Party or any other Person, shall be deemed to violate the rights of Bank under the Code. Such financing statements may indicate the Collateral as "all assets of the Debtor" or words of similar effect, or as being of an equal or lesser scope, or with greater detail, all in Bank's discretion.
- 4.4 <u>Intellectual Property.</u> Whenever any Loan Party, either by itself or through any agent, employee, licensee or designee, shall file an application for the registration of any Intellectual Property with the United States Patent and Trademark Office, the United States Copyright Office or any similar office or agency in any other country or any political subdivision thereof, such Loan Party shall report such filing to Bank concurrently with the next delivery of a Borrowing Base Certificate and/or Compliance Certificate.

Section 5 **Representations and Warranties**.

Each Loan Party represents and warrants as follows:

5.1 <u>Due Organization, Authorization; Power and Authority; Enforceability.</u>

(a) Such Loan Party and each of its Subsidiaries is duly organized or formed, validly existing and in good standing under the laws of the jurisdiction of its organization and is qualified and licensed to do business and is in good standing in each jurisdiction in which the conduct of its business or its ownership of property requires that it be qualified except where the failure to do so could not reasonably be expected to have a material adverse effect on Borrower's business. In connection with this Agreement, the Loan Parties have delivered to Bank a completed certificate signed by the Loan Parties, entitled "Perfection Certificate" (as updated by a notification to Bank in accordance with Section 7.2, the "Perfection Certificate"). Each Loan Party represents and warrants to Bank that (i) such Loan Party's exact legal name is that indicated on the Perfection Certificate and on the signature page hereof;

- 24 -

- (ii) such Loan Party is an organization of the type and is organized in the jurisdiction set forth in the Perfection Certificate; (iii) the Perfection Certificate accurately sets forth the Loan Parties' organizational identification number or accurately states that such Loan Party has none; (iv) the Perfection Certificate accurately sets forth the Loan Parties' place of business, or, if more than one, its chief executive office as well as each such Loan Party's mailing address (if different than its chief executive office); (v) no Loan Party (and each of its predecessors) has, in the past five (5) years, changed its jurisdiction of formation, organizational structure or type, or any organizational number assigned by its jurisdiction; and (vi) all other information set forth on the Perfection Certificate pertaining to the Loan Parties and their Subsidiaries is accurate and complete in all material respects (it being understood and agreed that the Loan Parties may from time to time update certain information in the Perfection Certificate after the Effective Date to the extent permitted by one (1) or more specific provisions in this Agreement and the Perfection Certificate shall be deemed to be updated to the extent such notice is provided to Bank of such permitted update). If any Loan Party is not now a Registered Organization but later becomes one, such Loan Party shall promptly notify Bank of such occurrence and provide Bank with such Loan Party's organizational identification number.
- (b) The execution, delivery and performance by each Loan Party of the Loan Documents have been duly authorized by such Loan Party, and do not (i) conflict with any of such Loan Party's organizational documents, (ii) contravene, conflict with, constitute a default under or violate any material Requirement of Law, (iii) contravene, conflict or violate any applicable order, writ, judgment, injunction, decree, determination or award of any Governmental Authority by which such Loan Party or any of its Subsidiaries or any of their property or assets may be bound, (iv) require any action by, filing, registration, or qualification with, or Governmental Approval from, any Governmental Authority (except such Governmental Approvals which have already been obtained and are in full force and effect and filings necessary to perfect Liens granted under the Loan Documents), or (v) conflict with, contravene, constitute a default or breach under, or result in or permit the termination or acceleration of, any material agreement by which Borrower is bound. No Loan Party is party to any agreement, instrument, document or arrangement that imposes any burdensome obligations. No Loan Party is in default under any agreement to which it is a party or by which it is bound in which the default could reasonably be expected to have a material adverse effect on such Loan Party's business.
- (c) This Agreement has been, and each other Loan Document, when delivered hereunder, will have been, duly executed and delivered by each Loan Party. This Agreement constitutes, and each other Loan Document when so delivered will constitute, a legal, valid and binding obligation of such Loan Party, enforceable against it in accordance with its terms.

5.2 <u>Property</u>.

(a) Each Loan Party has good title to, rights in, and the power to transfer each item of the Collateral upon which it purports to grant a Lien hereunder, free and clear of any and all Liens except Permitted Liens. No Loan Party has Collateral Accounts at or with any bank or financial

institution other than Bank or Bank's Affiliates except for (i) the Collateral Accounts described in the Perfection Certificate and which such Loan Party has taken or will take such actions as are necessary to give Bank a perfected security interest therein, pursuant to and to the extent required by Section 6.8(b) and (ii) other Collateral Accounts created after the Effective Date to the extent permitted under this Agreement and disclosed to Bank that are subject to Control Agreements except to the extent not required pursuant to Section 6.8(b).

(b) The Accounts are bona fide, existing obligations of the Account Debtors.

- 25 -

- (c) No third party bailee (such as a warehouse) is in possession of any Collateral except as otherwise provided in the Perfection Certificate or as permitted pursuant to Section 7.2. No Collateral shall be maintained at locations other than as provided in the Perfection Certificate or as permitted pursuant to Section 7.2.
- (d) All Inventory is in all material respects of good and marketable quality, free from material defects.
- (e) Each Loan Party and each of its Subsidiaries is the sole owner of the Intellectual Property which it owns or purports to own except for non-exclusive licenses granted to its customers in the ordinary course of business. No part of the Intellectual Property has been judged invalid or unenforceable, in whole or in part. Other than as disclosed to Bank, no claim has been made in writing that any part of the Intellectual Property owned by such Loan Party violates the rights of any third party except to the extent such claim could not reasonably be expected to have a material adverse effect on such Loan Party's business.
- (f) The security interest granted herein is and shall at all times continue to be a first priority perfected security interest in the Collateral (subject to Permitted Liens).
- (g) Except as noted on the Perfection Certificate or as otherwise disclosed to Bank in accordance with Section 6.10(b), neither Borrower nor any of its Subsidiaries is a party to, nor is it bound by, any Restricted License.

5.3 Accounts Receivable.

- (a) For each Account included in the most recent Borrowing Base Certificate, on the date each Advance is requested and made, such Account shall be an Eligible Account.
- (b) All statements made and all unpaid balances appearing in all invoices, instruments and other documents evidencing the Eligible Accounts are and shall be true and correct and all such invoices, instruments and other documents, and all of Borrower's Books are genuine and in all respects what they purport to be. Whether or not an Event of Default has occurred and is continuing, Bank may notify any Account Debtor owing Borrower money of Bank's security interest in such funds and verify the amount of such Eligible Account. All sales and other transactions underlying or giving rise to each Eligible Account shall comply in all material respects with all Applicable Law. Borrower has no knowledge of any actual or imminent Insolvency Proceeding of any Account Debtor whose accounts are Eligible Accounts in any Borrowing Base Certificate. To the best of Borrower's knowledge, all signatures and endorsements on all documents, instruments, and agreements relating to all Eligible Accounts are genuine, and all such documents, instruments and agreements are legally enforceable in accordance with their terms.
- 5.4 <u>Litigation</u>. There are no actions or proceedings pending or, to the knowledge of any Responsible Officer, threatened in writing by or against any Loan Party or any of their respective Subsidiaries that could reasonably be expected to result in liability involving more than, individually or in the aggregate, One Hundred Thousand Dollars (\$100,000.00).

5.5 <u>Financial Statements; No Default.</u>

(a) All consolidated financial statements related to Borrower and its Subsidiaries that Bank has received from Borrower fairly present in all material respects Borrower's

- 26 -

consolidated financial condition as of the date thereof and Borrower's consolidated results of operations for the period then ended.

(b) The Audited Financial Statements (i) were prepared in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein; (ii) fairly present the financial condition of Borrower and its Subsidiaries as of the date thereof and their results of operations, cash flows and changes in shareholders' equity for the period covered thereby in accordance with GAAP consistently applied throughout the period covered thereby, except as

otherwise expressly noted therein; and (iii) show all material indebtedness and other liabilities, direct or contingent, of Borrower and its Subsidiaries as of the date thereof, including liabilities for taxes, material commitments and Indebtedness.

- (c) No Default or Event of Default has occurred and is continuing or would result from the consummation of the transactions contemplated by this Agreement or any other Loan Document.
- 5.6 <u>Solvency.</u> The present fair salable value of each of Borrower's (individually) and the Loan Parties' (taken as a whole) consolidated assets (including goodwill minus disposition costs) exceeds the sum of Borrower's (individually) or the Loan Parties' (taken as a whole), as applicable, consolidated liabilities; the capital of each of Borrower (individually) and the Loan Parties (taken as a whole), is not unreasonably small in relation to its or their business, as applicable, as currently contemplated after the transactions in this Agreement; and each of Borrower (individually) and the Loan Parties (taken as a whole) are able to pay its or their debts (including trade debts), as applicable, as they mature and has not incurred or intends to incur debts including current obligations beyond its or their ability, as applicable, to pay such debts as they become due in the ordinary course of business.

5.7 <u>Regulatory Compliance</u>.

- (a) Each Loan Party and each of its respective Subsidiaries (and with respect to any plan subject to Title IV of ERISA, each of the Loan Parties' and their respective Subsidiaries' ERISA Affiliates) has met the minimum funding requirements of ERISA applicable to it with respect to any employee benefit plans subject to ERISA, and no event has occurred or is reasonably expected to occur with respect to any such plan that would result in any Loan Party or any Subsidiary incurring any material liability under ERISA. No Loan Party nor any of their respective Subsidiaries is an entity deemed to hold "plan assets" (within the meaning of the Plan Asset Regulations), and provided the Credit Extensions are not funded with "plan assets" the making of any Credit Extension hereunder will not give rise to a non-exempt prohibited transaction under Section 406 of ERISA or Section 4975 of the Code.
- (b) No Loan Party is an "investment company" or a company "controlled" by an "investment company" under the Investment Company Act of 1940, as amended. Borrower is not engaged as one of its important activities in extending credit for margin stock (under Regulations X, T and U of the Federal Reserve Board of Governors).
- (c) As of the date hereof, there are no strikes, lockouts or slowdowns against any Loan Party or any of their respective Subsidiaries pending or, to the knowledge of such Loan Party, threatened in writing. Each Loan Party and each of their respective Subsidiaries has complied in all material respects with all the provisions of the Federal Fair Labor Standards Act.
- (d) Each Loan Party and each of their respective Subsidiaries (i) has complied in all material respects with all Requirements of Law, and (ii) has not violated any Requirements of Law the violation of which could reasonably be expected to have a material adverse

- 27 -

effect on its business. None of any Loan Party's or any of their respective Subsidiaries' properties or assets has been used by such Loan Party or any of its respective Subsidiaries or, by previous Persons, in disposing, producing, storing, treating, or transporting any hazardous substance other than legally. Each Loan Party and each of its Subsidiaries has obtained all consents, approvals and authorizations of, made all declarations or filings with, and given all notices to, all Governmental Authorities that are necessary to continue its business as currently conducted, except where the failure to do so could not reasonably be expected to have a material adverse effect on such business.

5.8 <u>Subsidiaries; Investments</u>. No Loan Party nor any of its respective Subsidiaries own any stock, partnership, or other ownership interest or other equity securities except for Permitted Investments.

5.9 <u>Tax Returns and Payments; Pension Contributions.</u>

- (a) Each Loan Party and each of their respective Subsidiaries has timely filed all required material tax returns and reports, and each Loan Party and each of their respective Subsidiaries has timely paid all material foreign, federal, state and local taxes, assessments, deposits and contributions owed by such Loan Party or such Subsidiary except (i) with respect to any Subsidiary, where failure to do so could not reasonably be expected to result in a material adverse effect on such Subsidiary's business or (ii) to the extent such taxes are being contested in good faith by appropriate proceedings promptly instituted and diligently conducted, so long as such reserve or other appropriate provision, if any, as shall be required in conformity with GAAP shall have been made therefor.
- (b) The Loan Parties are unaware of any claims or adjustments proposed for any of such Loan Party's or any of its respective Subsidiaries' prior tax years which could result in additional taxes becoming due and payable by such Loan Party or such Subsidiary in excess of One Hundred Thousand Dollars (\$100,000.00) in the aggregate. Each Loan Party and each of its Subsidiaries (and with respect to any plan subject to Title IV of ERISA, each of the Loan Parties' and their respective

Subsidiaries' ERISA Affiliates) has paid all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms, and no Loan Party, nor any Subsidiary nor, with respect to any such plan subject to Title IV of ERISA, any ERISA Affiliate, has withdrawn from participation in, and has not permitted partial or complete termination of, or permitted the occurrence of any other event with respect to, any such plan which could reasonably be expected to result in any liability of such Loan Party, any of its respective Subsidiaries, or any ERISA Affiliates, as applicable, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency, except where failure to pay such amounts or such withdrawal, termination or liability could not reasonably be expected to result in a material adverse effect on such Loan Party's, Subsidiary's or ERISA Affiliate's business.

- 5.10 <u>Use of Proceeds</u>. Borrower shall use the proceeds of the Credit Extensions solely as working capital, for other general corporate purposes and not for personal, family, household or agricultural purposes.
- 5.11 <u>Full Disclosure</u>. No written representation, warranty or other statement of any Loan Party or any of their respective Subsidiaries in any certificate or written statement given to Bank, as of the date such representation, warranty, or other statement was made, taken together with all such written certificates and written statements given to Bank, contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements contained in the certificates or statements not misleading (it being recognized by Bank that the projections and forecasts provided by the Loan Parties in good faith and based upon reasonable assumptions are not viewed as facts and that actual

- 28 -

results during the period or periods covered by such projections and forecasts may differ from the projected or forecasted results).

- 5.12 <u>Insurance</u>. Borrower believes that the insurance maintained by or on behalf of the Loan Parties and their Subsidiaries is adequate and is customary for companies engaged in the same or similar businesses operating in the same or similar locations. As of the date hereof, all premiums in respect of such insurance have been paid.
- 5.13 Sanctions. No Loan Party, any of their respective Subsidiaries, any director or officer, or any employee, agent, or Affiliate, of a Loan Party or any of their respective Subsidiaries is a Person that is, or is owned or controlled by Persons that are, (a) the target of any sanctions administered or enforced by the US Department of the Treasury's Office of Foreign Assets Control, the US Department of State, the United Nations Security Council, the European Union, His Majesty's Treasury or the Hong Kong Monetary Authority (collectively, "Sanctions"), or (b) located, organized or resident in a country or territory that is the target of Sanctions, including currently, the so-called Luhansk People's Republic, the so-called Donetsk People's Republic, and the Crimea region of Ukraine, Cuba, Iran, North Korea and Svria
- 5.14 Anti-Bribery Laws. No Loan Party or any of their respective Subsidiaries nor to the knowledge of such Loan Party, any director, officer, agent, employee, Affiliate or other person acting on behalf of the Loan Parties or any of their respective Subsidiaries is aware of or has taken any action, directly or indirectly, that would result in a violation by such persons of any applicable anti-bribery law, including but not limited to, the United Kingdom Bribery Act 2010 (the "UK Bribery Act") and the U.S. Foreign Corrupt Practices Act of 1977 (the "FCPA"). Furthermore, each Loan Party and, to the knowledge of such Loan Party, its Affiliates, have conducted their businesses in compliance with the UK Bribery Act, the FCPA and similar laws, rules or regulations and have instituted and maintain policies and procedures designed to ensure, and which are reasonably expected to continue to ensure, continued compliance therewith.
- 5.15 <u>Affected Financial Institutions.</u> No Loan Party is an Affected Financial Institution.
- 5.16 <u>Beneficial Ownership Certificate</u>. As of the Effective Date, the information included in the Beneficial Ownership Certification, if applicable, is true and correct in all respects.
- 5.17 <u>Definition of "Knowledge."</u> For purposes of the Loan Documents, whenever a representation or warranty is made to a Loan Party's knowledge or awareness, to the "best of" Borrower's knowledge, or with a similar qualification, knowledge or awareness means the actual knowledge, after reasonable investigation, of any Responsible Officer.
- 5.18 <u>Healthcare Permits</u>. (a) Borrower and each Subsidiary of Borrower, has obtained all necessary Healthcare Permits and other rights from, and have made all necessary declarations and filings with, all applicable Governmental Authorities, all self-regulatory authorities and all courts and other tribunals necessary to engage in the management and/or operation of their respective business; (b) each such Healthcare Permit is valid and in full force and effect, and Borrower and each Subsidiary of Borrower, is in compliance with the terms and conditions of all such Healthcare Permits except where failure to be in such compliance or for a Healthcare Permit to be valid and in full force and effect could not reasonably be expected to have material adverse effect on Borrower's operations or business; and (c) neither Borrower nor any Subsidiary of Borrower has received notice from any Governmental Authority

5.19 Compliance with Healthcare Laws.

- (a) Borrower and each Subsidiary of Borrower is in compliance, in all material respects, with all applicable Healthcare Laws. Without limiting the generality of the foregoing, Borrower has not received written notice by a governmental authority of any material violation (or of any investigation, audit, or other proceeding involving allegations of any violation) of any Healthcare Laws, and no investigation, inspection, audit or other proceeding involving allegations of any material violation is, to the knowledge of Borrower, threatened in writing or contemplated.
- (b) To the knowledge of Borrower, no Borrower or Subsidiary of Borrower is in violation in any material respect of any law which is applicable to such Person its respective assets or the conduct of their respective businesses and neither Borrower or any Subsidiary of Borrower has been debarred or excluded from participation under a state or federal health care program, including any state or federal workers compensation program.
- (c) Neither Borrower or any Subsidiary of Borrower is a party to any corporate integrity agreements, deferred prosecution agreements, monitoring agreements, consent decrees, settlement orders or similar agreements with or imposed by any governmental authority.

Section 6 Affirmative Covenants.

Each Loan Party shall, and shall cause each of its Subsidiaries to, do all of the following:

6.1 Government Compliance.

- (a) Except as permitted by Section 7.3, maintain its legal existence and good standing in their respective jurisdictions of formation and maintain qualification in each jurisdiction in which the failure to so qualify could reasonably be expected to cause a Material Adverse Change. Each Loan Party shall comply, and have each Subsidiary comply, in all material respects, with all laws, ordinances and regulations to which it is subject.
- (b) Obtain all of the Governmental Approvals necessary for the performance by such Loan Party of its obligations under the Loan Documents and the grant of a security interest to Bank in the Collateral. Such Loan Party shall promptly provide copies of any such obtained Governmental Approvals to Bank.
- (c) Cause the operations and property of Borrower and each of its Subsidiaries to comply, in all material respects, with all applicable Healthcare Laws. Without limiting the foregoing, the operations and property of Borrower and each of its Subsidiaries shall comply with HIPAA in all material respects. Borrower has established and maintains a corporate compliance program that (i) addresses the material Requirements of Law, including all applicable Healthcare Laws, of Governmental Authorities having jurisdiction over its business and operations, and (ii) has been structured to account for the guidance issued by the U.S. Department of Health and Human Services regarding characteristics of effective corporate compliance programs.

6.2 <u>Financial Statements, Reports, Certificates</u>. Provide Bank with the following:

(a) a Borrowing Base Certificate (and any schedules related thereto) no later than thirty (30) days after the last day of each month and at the time of a request for an Advance, with such request for an Advance;

- 30 -

- (b) as soon as available, but no later than thirty (30) days after the last day of each month, monthly accounts receivable agings, aged by invoice date and monthly accounts payable agings, aged by invoice date;
- (c) as soon as available, but no later than thirty (30) days after the last day of each month, a company prepared consolidated balance sheet, income statement and cash flow statement covering Borrower's consolidated operations for such month certified by a Responsible Officer of Borrower and in a form reasonably acceptable to Bank;
- (d) as soon as available, but no later than thirty (30) days after the last day of each month, a duly completed Compliance Certificate signed by a Responsible Officer of Borrower, certifying that as of the end of such month, the Loan Parties were in full compliance with all of the terms and conditions of this Agreement and the other Loan Documents, and setting forth calculations showing

compliance with the financial covenants set forth in this Agreement and such other information as Bank may reasonably request;

- as soon as available, but no later than sixty (60) days after the last day of (e) each fiscal year of Borrower and contemporaneously with any amendments thereto, (A) annual operating budgets (including income statements, balance sheets and cash flow statements, by month) for the then current fiscal year of Borrower, and (B) annual financial projections for such fiscal year (on a monthly basis) as approved by the Board, together with any related business forecasts used in the preparation of such annual financial projections;
- as soon as available, and in any event within one hundred eighty (180) days following the end of each fiscal year of Borrower (commencing with Borrower's fiscal year ending December 31, 2023), audited consolidated financial statements prepared under GAAP, consistently applied, together with an unqualified opinion on the financial statements from an independent certified public accounting firm reasonably acceptable to Bank; provided, however, that Borrower shall not be required to deliver such materials if the Board does not require audited annual financial statements for any fiscal year of Borrower, in which case Borrower shall instead be required to deliver for such fiscal year, as soon as available, but no later than one hundred eighty (180) days after the last day of such fiscal year, company prepared consolidated financial statements for such fiscal year, prepared under GAAP, consistently applied, certified by a Responsible Officer and in a form acceptable to Bank;
- at least annually, and within thirty (30) days of any material change in the (g) fully-diluted shares of Borrower, a current capitalization table of Borrower, in form and detail acceptable to Bank;
 - (h) [intentionally omitted];
- in the event that Borrower or any Subsidiary becomes subject to the reporting requirements under the Exchange Act, within five (5) days of filing, copies of all periodic and other reports, proxy statements and other materials filed by Borrower with the SEC, any Governmental Authority succeeding to any or all of the functions of the SEC or with any national securities exchange, or distributed to its shareholders, as the case may be. Documents required to be delivered pursuant to the terms hereof (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date on which Borrower posts such documents, or provides a link thereto, on Borrower's website on the internet at Borrower's website address; provided, however, Borrower shall promptly notify Bank in writing (which may be by electronic mail) of the posting of any such documents;

- within ten (10) Business Days of delivery, copies of all statements, reports and notices made available to such Loan Party's security holders or to any holders of Subordinated Debt:
- promptly, and in any event, within three (3) Business Days, report of the occurrence of any Default or Event of Default;
- promptly, and in any event, within five (5) Business Days, report of (i) any adverse finding in respect of any action or proceeding set forth in the Perfection Certificate, (ii) any legal actions pending or threatened in writing against any Loan Party or any of their respective Subsidiaries that could reasonably be expected to result in damages or costs to the Loan Parties or any of their respective Subsidiaries of, individually or in the aggregate, One Hundred Thousand Dollars (\$100,000.00) or more, (iii) any Acquisitions or (iv) any matter that has resulted or could reasonably be expected to result in a Material Adverse Change;
- promptly following any request therefor, information and documentation (m) reasonably requested by Bank for purposes of compliance with applicable "know your customer" and anti-money-laundering rules and regulations, including, without limitation, the Patriot Act and the Beneficial Ownership Regulation;
- prompt, and in any event, within five (5) Business Days, report of any departure of any Key Person departing or ceasing to be employed by Borrower or of any Key Person ceasing to be involved in the day to day operations of the Loan Parties or to hold an executive office at least equal in seniority and responsibility to such Person's present office as of the Effective Date;
- within thirty (30) days after the last day of each month, a detailed cash report, including monthly bank statements, setting forth each Loan Party's and each of their Subsidiaries' accounts maintained outside of Bank and Bank's Affiliates, and the balances maintained in such accounts as of the last day of such month, in a form reasonably acceptable to Bank;
 - other financial information reasonably requested by Bank promptly after

- (q) prompt notice of the creation or acquisition of any Subsidiary, including if such Subsidiary is a Foreign Subsidiary; and
- (r) at least five (5) Business Days' (or such shorter period as may be agreed to by Bank) prior written notice of any sale or issuance of any stock of Borrower which will result in any Person owning, directly or indirectly, ten percent (10.0%) or more of the outstanding voting stock of Borrower on a fully diluted basis, including the purchasers of such stock, any "know your customer" information required by Bank, the terms of such sale or issuance and the total sale or issuance proceeds to be received by Borrower.
- 6.3 <u>Inventory; Returns.</u> Keep all Inventory in good and marketable condition, free from material defects. Returns and allowances (a) between any Loan Party and its respective Account Debtors and (b) between any Subsidiary of any Loan Party and its respective Account Debtors, in each case, shall follow the customary practices of such Loan Party and such Subsidiary, respectively, as they exist on the Effective Date. Borrower shall promptly notify Bank of all returns, recoveries, disputes and claims that involve more than One Hundred Thousand Dollars (\$100,000.00).

- 32 -

- 6.4 <u>Maintenance of Properties</u>. (a) Maintain, preserve and protect all of its material properties and equipment necessary in the operation of its business in good working order and condition, ordinary wear and tear excepted; and (b) make all necessary repairs thereto and renewals and replacements thereof except where the failure to do so could not reasonably be expected to have a Material Adverse Change.
- 6.5 <u>Payment of Obligations; Taxes; Pensions.</u> Timely file all required material tax returns and reports and timely pay all its material obligations and liabilities, including, without limitation, foreign, federal, state and local taxes, assessments, deposits and contributions owed by the Loan Parties and each of their respective Subsidiaries, except for deferred payment of any taxes contested pursuant to the terms of Section 5.9 hereof, and shall deliver to Bank, upon reasonable request, appropriate certificates attesting to such payments, and pay all amounts necessary to fund all present pension, profit sharing and deferred compensation plans (including, with respect to any such plan subject to Title IV of ERISA, evidence that each ERISA Affiliate has paid such amounts) in accordance with their terms.

6.6 Access to Collateral; Books and Records.

- (a) At reasonable times, on five (5) Business Days' written notice (provided no notice is required if an Event of Default has occurred and is continuing), provide Bank, or its agents, the right, during normal business hours (except if an Event of Default has occurred and is continuing), to inspect the Collateral and the right to audit and copy the Books. Such inspections and audits shall be conducted as frequently as Bank determines in its sole discretion that conditions warrant with reasonable efforts to minimize disruption to such Loan Party's business. The foregoing inspections and audits shall be at Borrower's expense no more often than once in any year, unless an Event of Default has occurred and is continuing.
- (b) At reasonable times, on five (5) Business Days' written notice (provided no notice is required if an Event of Default has occurred and is continuing), provide Bank, or its agents, the right, during normal business hours (except if an Event of Default has occurred and is continuing), to discuss its financial matters with the independent auditors of Borrower and its Subsidiaries (and each Loan Party hereby authorizes all such independent auditors to discuss those financial matters with Bank or any representative, agent, or advisor thereof). The foregoing shall be conducted at Borrower's expense and no more often than once every twelve (12) months, unless an Event of Default has occurred and is continuing, in which case such discussions shall occur as often as Bank shall reasonably determine is necessary.
- (c) Maintain proper books of record and account in which full, true and correct entries in conformity with GAAP consistently applied shall be made of all financial transactions and matters involving the assets and business of such Loan Party or such Subsidiary, as the case may be.

6.7 <u>Insurance</u>.

(a) Keep its business and the Collateral insured for risks and in amounts standard for similarly situated companies in such Loan Party's industry, stage of development and location and as Bank may reasonably request. Insurance policies shall be in a form, with financially sound and reputable insurance companies that are not Affiliates of any Loan Party, and in amounts that are reasonably satisfactory to Bank. All property policies that name any Loan Party as an insured party shall have a lender's loss payable endorsement showing Bank as lender loss payee. All liability policies that name any Loan Party as an insured party shall show, or have endorsements showing, Bank as an additional insured. Bank shall be named as lender loss payee and/or additional insured with respect to any such insurance providing coverage in respect of any Collateral.

- (b) Ensure that proceeds payable under any property policy are, at Bank's option, payable to Bank on account of the Obligations, provided that, so long as no Event of Default has occurred and is continuing, such Loan Party may apply the proceeds of any property policy toward the replacement or repair of destroyed or damaged property; provided that any such replaced or repaired property shall be (A) of equal or like value as the replaced or repaired property and (B) deemed Collateral in which Bank has been granted a first priority security interest to the extent the destroyed or damaged property consists of Collateral.
- (c) At Bank's request, the Loan Parties shall deliver certificates of insurance and evidence of all premium payments. Each provider of any such insurance required under this Section 6.7 shall agree, by endorsement upon the policy or policies issued by it or by independent instruments furnished to Bank, that it will give Bank thirty (30) days' prior written notice before any such policy or policies shall be altered or canceled (with the exception of cancellation due to non-payment of premium, which will be ten (10) days' prior written notice). If any Loan Party fails to obtain insurance as required under this Section 6.7 or to pay any amount or furnish any required proof of payment to third persons and Bank, Bank may make all or part of such payment or obtain such certificates of insurance policies required in this Section 6.7, and take any action under the policies Bank deems reasonably prudent.

6.8 Operating Accounts.

- Each Loan Party and each of their Subsidiaries shall maintain (i) at least seventy-five percent (75.0%) of its depository and operating accounts, Cash Equivalents, and excess cash and (ii) upon the expiration of the Transition Period, the Collections Account (as defined below), with Bank or Bank's Affiliates. In addition to the foregoing, each Loan Party and each Subsidiary shall obtain any business credit card and any letter of credit exclusively from Bank or an Affiliate of Bank. In addition, upon the expiration of the Transition Period, Borrower shall direct Account Debtors to deliver and transmit all proceeds of Accounts into a collections account of Borrower maintained with an Affiliate of Bank determined by Bank (the "Collections Account") and all available funds in the Collections Account shall be (i) when a Streamline Period is not in effect, (A) if there are Obligations under the Revolving Line outstanding, applied to reduce the Obligations under the Revolving Line within seven (7) Business Days of receipt thereof or (B) if there are no Obligations outstanding under the Revolving Line and no Event of Default has occurred, transferred to an account of Borrower maintained with Bank or an Affiliate of Bank as determined by Bank within seven (7) Business Days of receipt thereof (or such shorter timeframe with respect to (A) and (B) as Bank may agree to in writing in its sole discretion on a case-by-case basis upon receipt of a written request from Borrower requesting that such proceeds be applied or remitted as the case may be as of an earlier date), or (ii) when a Streamline Period is in effect, transferred on a daily basis to an account of Borrower maintained with Bank or an Affiliate of Bank.
- (b) Provide Bank written notice within five (5) Business Days after establishing any Collateral Account of any Loan Party at or with any bank or financial institution other than Bank or Bank's Affiliates. For each Collateral Account that any Loan Party at any time maintains, such Loan Party shall cause the applicable bank or financial institution (other than Bank, but including Bank's Affiliates) at or with which any Collateral Account is maintained to execute and deliver within thirty (30) days after opening such Collateral Account a Control Agreement or other appropriate instrument with respect to such Collateral Account to perfect Bank's Lien in such Collateral Account in accordance with the terms hereunder which Control Agreement may not be terminated without the prior written consent of Bank. The provisions of the previous sentence shall not apply to deposit accounts exclusively used for payroll, payroll taxes, and other employee wage and benefit payments to or for the benefit of such Loan Party's employees and identified to Bank by such Loan Party as such (such deposit accounts, "Excluded Accounts").

- 34 -

6.9 <u>Financial Covenant – Minimum Adjusted Quick Ratio</u>. Maintain at all times, to be tested as of the last day of each month, an Adjusted Quick Ratio of at least 1.50 to 1.00.

6.10 <u>Protection of Intellectual Property Rights</u>.

- (a) (i) Protect, defend and maintain the validity and enforceability of its Intellectual Property; (ii) promptly advise Bank in writing of material infringements or any other event that could reasonably be expected to materially and adversely affect the value of its Intellectual Property that is material to such Loan Party's business; and (iii) not allow any Intellectual Property that is material to such Loan Party's business (as determined in good faith by such Loan Party) to be abandoned, forfeited or dedicated to the public without Bank's written consent.
- (b) Provide written notice to Bank within ten (10) days of entering or becoming bound by any Restricted License (other than over-the-counter software that is commercially available to the public). At Bank's request, Borrower shall use commercially reasonable efforts to obtain the consent of, or waiver by, any person whose consent or waiver is necessary for (i) any Restricted License of any Loan Party to be deemed "Collateral" and for Bank to have a security interest in it that might otherwise be restricted or prohibited by law or by the terms of any such Restricted License, whether now existing or entered into in the future, and (ii) Bank to have the ability in the event of a liquidation of

any Collateral to dispose of such Collateral in accordance with Bank's rights and remedies under this Agreement and the other Loan Documents.

- 6.11 <u>Litigation Cooperation</u>. From the date hereof and continuing through the termination of this Agreement, make available to Bank, without expense to Bank, each Loan Party and its respective officers, employees and agents and the Books, to the extent that Bank may deem them reasonably necessary to prosecute or defend any third-party suit or proceeding instituted by or against Bank with respect to any Collateral or relating to any Loan Party.
- Formation or Acquisition of Subsidiaries. Notwithstanding and without limiting the negative covenants contained in Sections 7.3 and 7.7 hereof, at the time that any Loan Party forms any direct or indirect Subsidiary or acquires any direct or indirect Subsidiary after the Effective Date (including, without limitation, pursuant to a Division), such Loan Party shall or shall cause such Subsidiary to provide to Bank, within thirty (30) days after such acquisition or formation: (a) a joinder to this Agreement to become a co-borrower hereunder or a Guarantee Agreement to become a Guarantor hereunder (as determined by Bank in its sole discretion) together with such appropriate financing statements and/or Control Agreements, all in form and substance satisfactory to Bank (including being sufficient to grant Bank a first priority Lien (subject to Permitted Liens) in and to the assets of such Subsidiary); (b) appropriate certificates and powers, proxies and financing statements, pledging all of the direct ownership interest of any Loan Party in such Subsidiary in form and substance satisfactory to Bank; and (c) all other documentation in form and substance reasonably satisfactory to Bank, including one (1) or more perfection certificates, secretary certificates, and opinions of counsel satisfactory to Bank, which in its opinion is appropriate with respect to the execution and delivery of the applicable documentation referred to above. Any document, agreement, or instrument executed or issued pursuant to this Section 6.12 shall be a Loan Document.
- 6.13 <u>Use of Proceeds</u>. Use the proceeds of the Credit Extensions as working capital, and for other general corporate purposes not in contravention of any law or of any Loan Document.
- 6.14 <u>Further Assurances</u>. Execute any further instruments and take further action as Bank reasonably requests to perfect or continue Bank's Lien in the Collateral, to ensure the Obligations are guaranteed by each Subsidiary of Borrower or to effect the purposes of this Agreement and the other

- 35 -

Loan Documents. Deliver to Bank, within five (5) Business Days after the same are sent or received, copies of all correspondence, reports, documents and other filings with any Governmental Authority regarding compliance with or maintenance of Governmental Approvals or Requirements of Law that could reasonably be expected to have a material adverse effect on any of the Governmental Approvals or otherwise on the operations of such Loan Party or any of its respective Subsidiaries.

6.15 <u>Post-Closing Conditions</u>. Borrower shall deliver to Bank:

- (a) Within thirty (30) days of the Effective Date, duly executed Control Agreements with HSBC Bank USA, N.A. with respect to (i) the Collections Account and (ii) Borrower's account with HSBC Bank USA, N.A. ending ****; and
- (b) If any account of Borrower remains open at KeyBank, N.A., within thirty (30) days of the Effective Date (or such later date, as agreed to by Bank in writing in its sole and absolute discretion), a duly executed Control Agreement with Key Bank, N.A., in form and substance acceptable to Bank.

Section 7 **Negative Covenants**.

No Loan Party shall, nor shall it permit any of its Subsidiaries to, do any of the following without Bank's prior written consent:

Dispositions. Convey, sell, lease, transfer, assign, or otherwise dispose of (including, without limitation, pursuant to a Division) (collectively, "Transfer") all or any part of its business or property, except for Transfers (a) of Inventory in the ordinary course of business; (b) of wornout or obsolete Equipment that is, in the reasonable judgment of Borrower, no longer economically practicable to maintain or useful in the ordinary course of business of Borrower or any of its Subsidiaries, including, for the avoidance of doubt, exchanges of worn-out or obsolete Equipment for other Equipment; (c) consisting of Permitted Liens and Permitted Investments; (d) consisting of Borrower's or its Subsidiaries' use or transfer of money or Cash Equivalents in a manner that is not prohibited by the terms of this Agreement or the other Loan Documents; (e) from any Subsidiary that is not a Loan Party to Borrower or any Subsidiary; (f) from any Loan Party to another Loan Party, and (g) other Transfers not to exceed One Hundred Thousand Dollars (\$100,000.00) in the aggregate in any twelve (12) month period; provided that in no event shall any Loan Party Transfer (i) Intellectual Property to a Subsidiary that is not a Loan Party, or (ii) any customer contracts that generate receivables or revenue.

- (a) (i) Engage in any business other than the businesses currently engaged in by the Loan Parties and their Subsidiaries or that are reasonably similar, related or incidental thereto; or (ii) change the date on which its fiscal year ends.
- (b) No Loan Party shall, without at least thirty (30) days prior written notice to Bank: (i) add any new locations, (ii) deliver any material Collateral to any third party bailee, (iii) change its jurisdiction of organization, (iv) change its organizational type, (v) change its legal name, or (vi) change any organizational number (if any) assigned by its jurisdiction of organization. Each notice under this Section 7.2(b) shall be deemed to update the Perfection Certificate to include such information.
- (c) If Borrower intends to add any new offices or business locations, including warehouses, containing in excess of One Hundred Thousand Dollars (\$100,000.00) of Borrower's assets or property, then Borrower shall use commercially reasonable efforts to cause the landlord of any such new offices or business locations, including warehouses, to execute and deliver a

- 36 -

landlord consent in form and substance satisfactory to Bank. If Borrower intends to deliver any portion of the Collateral valued, individually or in the aggregate, in excess of One Hundred Thousand Dollars (\$100,000.00) to a bailee, and Bank and such bailee are not already parties to a bailee agreement governing both the Collateral and the location to which Borrower intends to deliver the Collateral, then Borrower shall use commercially reasonable efforts to cause such bailee to execute and deliver a bailee agreement in form and substance satisfactory to Bank.

- 7.3 <u>Fundamental Changes; Acquisitions</u>. Dissolve, liquidate, merge or consolidate with any other Person, make any Acquisition, or permit any of its Subsidiaries to make any Acquisition (including, without limitation, by the formation of any Subsidiary or pursuant to a Division); provided that any Subsidiary may dissolve or liquidate its assets in favor of a Loan Party or, in the case of any Subsidiary that is not a Loan Party, in favor of a Loan Party or other Subsidiary of Borrower.
- 7.4 <u>Indebtedness</u>. Create, incur, assume, or be liable for any Indebtedness other than Permitted Indebtedness.
- 7.5 <u>Encumbrances; Negative Pledge</u>. (a) Create, incur, allow, or suffer any Lien on any of its property, or assign or convey any right to receive income, including the sale of any Accounts or Contracts, except for Permitted Liens, (b) permit any Collateral not to be subject to the first priority security interest granted herein (subject to Permitted Liens), or (c) enter into any agreement, document, instrument or other arrangement (except with or in favor of Bank) with any Person which directly or indirectly prohibits or has the effect of prohibiting such Loan Party or any of its Subsidiaries from assigning, mortgaging, pledging, granting a security interest in or upon, or encumbering any of such Loan Party's or such Subsidiary's Intellectual Property in favor of Bank.
- 7.6 <u>Maintenance of Collateral Accounts</u>. Maintain any Collateral Account except pursuant to the terms of Section 6.8(b) hereof.
- 7.7 <u>Distributions; Investments.</u> (a) Pay any dividends or make any distribution or payment or redeem, retire or purchase any capital stock or; or (b) directly or indirectly make any Investment (including, without limitation, by the formation of any Subsidiary) other than Permitted Investments.
- 7.8 Transactions with Affiliates. Directly or indirectly enter into or permit to exist any material transaction with any Affiliate of such Loan Party, except for (a) transactions permitted by Section 7.1 and Section 7.7, (b) transactions by and among the Loan Parties permitted hereunder, (c) reasonable and customary compensation arrangements approved by the Board, and (d) transactions that are in the ordinary course of such Loan Party's business, upon fair and reasonable terms that are no less favorable to such Loan Party than would be obtained in an arm's length transaction with a non-affiliated Person.
- 7.9 <u>Subordinated Debt.</u> (a) Make or permit any payment on any Subordinated Debt, except under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject, or (b) amend any provision in any document relating to the Subordinated Debt which would increase the amount thereof, provide for earlier or greater principal, interest, or other payments thereon, or adversely affect the subordination thereof to Obligations owed to Bank, except as expressly permitted under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject.
- 7.10 <u>Compliance</u>. (a) Become an "investment company" or a company controlled by an "investment company", under the Investment Company Act of 1940, as amended, or undertake as one

of its important activities extending credit to purchase or carry margin stock (as defined in Regulation U of the Board of Governors of the Federal Reserve System), or use the proceeds of any Credit Extension for that purpose; (b) fail to meet the minimum funding requirements of ERISA, as applicable, permit a Reportable Event or Prohibited Transaction, each as defined in ERISA, to occur; (c) fail to comply with the Federal Fair Labor Standards Act or violate any other law or regulation, if the violation could reasonably be expected to have a material adverse effect on such Loan Party's business; or (d) withdraw from participation in, permit partial or complete termination of, or permit the occurrence of any other event with respect to, any present pension, profit sharing and deferred compensation plan (or permit any ERISA Affiliate to do the foregoing with respect to any such plan subject to Title IV of ERISA) which could reasonably be expected to result in any material liability of such Loan Party, including any material liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency.

- 7.11 Restrictive Agreements. Enter into or permit to exist any contractual obligation (other than this Agreement or any other Loan Document or any Bank Services Agreement) that (a) limits the ability (i) of any Subsidiary to pay any dividends or make any distribution or payment to, or redeem, retire or purchase any capital stock of, Borrower or to otherwise transfer property to or invest in Borrower, or (ii) of any Subsidiary to guarantee the Indebtedness of Borrower or grant a Lien on its assets in favor of Bank; or (b) requires the grant of a Lien (other than a Permitted Lien) to secure an obligation of such Person if a Lien is granted to secure another obligation of such Person.
- 7.12 <u>Sanctions</u>. Directly or indirectly, use the proceeds of the Credit Extensions, or lend, contribute or otherwise make available such proceeds to any Subsidiary, joint venture partner or other Person, (a) to fund any activities or business of or with any Person, or in any country or territory, that, at the time of such funding, is the target or subject of Sanctions or (b) in any other manner that would result in a violation of Sanctions by any Person (including any Person participating in the Credit Extensions, whether as issuing bank, lender, underwriter, advisor, investor or otherwise).
- 7.13 <u>Anti-Bribery.</u> Use, directly or indirectly, any part of the proceeds of any Credit Extension for any payments that could constitute a violation of any applicable anti-bribery law.

Section 8 **Events of Default**.

Any one of the following shall constitute an event of default (an "Event of Default") under this Agreement:

8.1 <u>Payment Default</u>. Any Loan Party fails to (a) make any payment of principal or interest on any Credit Extension when due, or (b) pay any other Obligations within three (3) Business Days after such Obligations are due and payable. During the cure period, the failure to make or pay any payment specified under clause (b) hereunder is not an Event of Default (but no Credit Extension will be made during the cure period);

8.2 <u>Covenant Default</u>.

- (a) Any Loan Party fails or neglects to perform any obligation in Sections 6.1(a) (with respect to maintaining a Loan Party's existence), 6.2, 6.5, 6.7, 6.8, 6.9, 6.10(b), 6.12, 6.13, 6.14 or violates any covenant in Section 7; or
- (b) Any Loan Party fails or neglects to perform, keep, or observe any other term, provision, condition, covenant or agreement contained in this Agreement or any other Loan Document or any Bank Services Agreement, and as to any default (other than those specified in Section 8.1 or Section 8.2(a)) under such other term, provision, condition, covenant or agreement that can be cured, has failed to cure the default within ten (10) days after the occurrence thereof; provided, however,

- 38 -

that if the default cannot by its nature be cured within the ten (10) day period or cannot after diligent attempts by such Loan Party be cured within such ten (10) day period, and such default is likely to be cured within a reasonable time, then such Loan Party shall have an additional period (which shall not in any case exceed thirty (30) days) to attempt to cure such default, and within such reasonable time period the failure to cure the default shall not be deemed an Event of Default (but no Credit Extensions shall be made during such cure period);

- 8.3 <u>Material Adverse Change</u>. A Material Adverse Change occurs;
- 8.4 <u>Attachment; Levy; Restraint on Business</u>.
- (a) (i) The service of process seeking to attach, by trustee or similar process, any funds of any Loan Party or of any Subsidiary, or (ii) a notice of lien or levy is filed against any of such Loan Party's assets by any Governmental Authority, and the same under subclauses (i) and (ii) hereof are not, within ten (10) days after the occurrence thereof, discharged or stayed (whether through the posting of a bond or otherwise); provided, however, no Credit Extensions shall be made during any ten (10) day cure period; or

- (b) (i) any material portion of any Loan Party's or any Subsidiary's assets is attached, seized, levied on, or comes into possession of a trustee or receiver, or (ii) any court order enjoins, restrains, or prevents such Loan Party or any Subsidiary from conducting all or any material part of its business:
- 8.5 <u>Insolvency.</u> (a) Any Loan Party or any of its Subsidiaries is unable to pay its debts (including trade debts) as they become due or otherwise becomes insolvent as described in Section 5.6; (b) Borrower or any of its Subsidiaries begins an Insolvency Proceeding; or (c) an Insolvency Proceeding is begun against any Loan Party or any of its Subsidiaries and is not dismissed or stayed within thirty (30) days (but no Credit Extensions shall be made while any of the conditions described in clause (a) exist and/or until any Insolvency Proceeding is dismissed);
- 8.6 Other Agreements. There is, under (a) any agreement to which any Loan Party or any Subsidiary is a party with a third party or parties, any default resulting in a right by such third party or parties, whether or not exercised, to accelerate the maturity of any Indebtedness in an amount individually or in the aggregate in excess of One Hundred Thousand Dollars (\$100,000.00), or (b) any breach or default by Borrower, any of Borrower's Subsidiaries, or Guarantor, the result of which could have a material adverse effect on Borrower's, any of Borrower's Subsidiaries', or any Guarantor's business or operations;
- 8.7 <u>Judgments; Penalties</u>. One or more fines, penalties or final judgments, orders or decrees for the payment of money in an amount, individually or in the aggregate, of at least One Hundred Thousand Dollars (\$100,000.00) (not covered by independent third-party insurance as to which liability has been accepted by such insurance carrier) shall be rendered against any Loan Party or any Subsidiary by any Governmental Authority, and the same are not, within ten (10) days after the entry, assessment or issuance thereof, discharged, satisfied, or paid, or after execution thereof, stayed or bonded pending appeal, or such judgments are not discharged prior to the expiration of any such stay (provided that no Credit Extensions will be made prior to the satisfaction, payment, discharge, stay, or bonding of such fine, penalty, judgment, order or decree);
- 8.8 <u>Misrepresentations</u>. Any Loan Party has made any representation, warranty, or other statement in any Loan Document, any Bank Services Agreement or in any writing delivered to Bank, in order to induce Bank to enter this Agreement, any Loan Document or any Bank Services

- 39 -

Agreement, and such representation, warranty, or other statement is incorrect in any material respect when made;

- 8.9 <u>Subordinated Debt.</u> Any subordination or intercreditor agreement with respect to any Subordinated Debt shall for any reason be revoked or invalidated or otherwise cease to be in full force and effect, any Person (other than Bank) shall be in breach thereof or contest in any manner the validity or enforceability thereof or deny that it has any further liability or obligation thereunder, or the Obligations shall for any reason be subordinated or shall not have the priority contemplated by this Agreement or any applicable subordination or intercreditor agreement;
- 8.10 <u>Loan Documents</u>. Any Loan Document, or any material provision thereof, at any time after its execution and delivery and for any reason other than as expressly permitted hereunder or thereunder or satisfaction in full of all Obligations, ceases to be in full force and effect; or any Loan Party or any other Person contests in writing the validity or enforceability of any Loan Document or any Bank Services Agreement or, in either case, any provision thereof; or any Loan Party denies in writing that it has any or further liability or obligation under any Loan Document, or purports in writing to revoke, terminate or rescind any Loan Document;
 - 8.11 <u>Change in Control</u>. The occurrence of a Change in Control; or
- 8.12 <u>Governmental Approvals</u>. Any Governmental Approval shall have been revoked, rescinded, suspended, modified in an adverse manner or not renewed that would reasonably be expected to cause a Material Adverse Change.
- 8.13 <u>Delisting</u>. After an initial public offering of Borrower's common stock on an exchange or market, such shares are delisted from such exchange or market because of Borrower's failure to comply with continued listing standards thereof or due to a voluntary delisting which results in such shares not being listed on such exchange or market.

Section 9 Bank's Rights and Remedies.

- 9.1 <u>Rights and Remedies</u>. Upon the occurrence and during the continuance of an Event of Default, Bank may, without notice or demand, do any or all of the following:
- (a) declare all Obligations to be immediately due and payable (but if an Event of Default described in Section 8.5 occurs all Obligations shall be automatically and immediately due and payable without any action by Bank);

- (b) stop advancing money or extending credit for Borrower's benefit under this Agreement or under any other agreement between Borrower and Bank;
- (c) demand that Borrower (i) deposit cash with Bank in an amount equal to at least (A) one hundred five percent (105.0%) of the aggregate face amount of any letters of credit denominated in Dollars remaining undrawn, and (B) one hundred fifteen percent (115.0%) of the Dollar Equivalent of the aggregate face amount of any letters of credit denominated in a Foreign Currency remaining undrawn (plus, in each case, all interest, fees, and costs due or estimated by Bank to become due in connection therewith), to secure all of the Obligations relating to such letters of credit, as collateral security for the repayment of any future drawings under such letters of credit, and Borrower shall forthwith deposit and pay such amounts, and (ii) pay in advance all letter of credit fees scheduled to be paid or payable over the remaining term of any letters of credit;

- 40 -

- (d) terminate any FX Contracts (it being understood and agreed that (i) Bank is not obligated to deliver the currency which Borrower has contracted to receive under any FX Contract, and Bank may cover its exposure for any FX Contracts by purchasing or selling currency in the interbank market as Bank deems appropriate; (ii) Borrower shall be liable for all losses, damages, costs, margin obligations and expenses incurred by Bank arising from Borrower's failure to satisfy its obligations under any FX Contract or the execution of any FX Contract; and (iii) Bank shall not be liable to Borrower for any gain in value of a FX Contract that Bank may obtain in covering Borrower's breach);
- (e) verify the amount of, demand payment of and performance under, and collect any Accounts and General Intangibles, settle or adjust disputes and claims directly with Account Debtors for amounts on terms and in any order that Bank considers advisable, and notify any Person owing Borrower money of Bank's security interest in such funds;
- (f) make any payments and do any acts it considers necessary or reasonable to protect or preserve the Collateral and/or its security interest in the Collateral. The Loan Parties shall assemble the Collateral if Bank requests and make it available as Bank designates. Bank may enter premises where the Collateral is located, take and maintain possession of any part of the Collateral, and pay, purchase, contest, or compromise any Lien which appears to be prior or superior to its security interest and pay all expenses incurred. Each Loan Party grants Bank a license to enter and occupy any of its premises, without charge, to exercise any of Bank's rights or remedies;
- (g) apply to the Obligations any (i) balances and deposits of any Loan Party it holds, or (ii) amount held by Bank owing to or for the credit or the account of the Loan Parties;
- (h) ship, reclaim, recover, store, finish, maintain, repair, prepare for sale, advertise for sale, and sell the Collateral. For use solely upon the occurrence and during the continuation of an Event of Default, Bank is hereby granted a non-exclusive, royalty-free license or other right to use, without charge, Borrower's labels, Patents, Copyrights, mask works, rights of use of any name, trade secrets, trade names, Trademarks, and advertising matter, or any similar property as it pertains to the Collateral, in completing production of, advertising for sale, and selling any Collateral and, in connection with Bank's exercise of its rights under this Section 9.1, each Loan Party's rights under all licenses and all franchise agreements inure to Bank's benefit;
- (i) place a "hold" on any account maintained with Bank and/or deliver a notice of exclusive control, any entitlement order, or other directions or instructions pursuant to any Control Agreement or similar agreements providing control of any Collateral (other than deposit accounts exclusively used for payroll, payroll taxes, and other employee wage and benefit payments to or for the benefit of the Loan Parties' employees and identified to Bank by Borrower as such);
 - (j) demand and receive possession of the Books; and
- (k) exercise all rights and remedies available to Bank under the Loan Documents or any Bank Services Agreement or at law or equity, including all remedies provided under the Code (including disposal of the Collateral pursuant to the terms thereof).
- 9.2 <u>Power of Attorney.</u> Each Loan Party hereby irrevocably appoints Bank as its lawful attorney-in-fact, exercisable upon the occurrence and during the continuance of an Event of Default, to: (a) endorse such Loan Party's name on any checks or other forms of payment or security; (b) sign such Loan Party's name on any invoice or bill of lading for any Account or drafts against Account Debtors; (c) settle and adjust disputes and claims about the Accounts directly with Account Debtors; (d) make, settle, and adjust all claims under such Loan Party's insurance policies; (e) pay, contest or settle

appoints Bank as its lawful attorney-in-fact to sign such Loan Party's name on any documents necessary to perfect or continue the perfection of Bank's security interest in the Collateral regardless of whether an Event of Default has occurred until all Obligations (other than inchoate indemnity obligations) have been satisfied in full and Bank is under no further obligation to make Credit Extensions hereunder. Bank's foregoing appointment as such Loan Party's attorney-in-fact, and all of Bank's rights and powers, coupled with an interest, are irrevocable until all Obligations have been fully repaid and performed and Bank's obligation to provide Credit Extensions terminates.

- 9.3 Protective Payments. If any Loan Party fails to obtain the insurance called for by Section 6.7 or fails to pay any premium thereon or fails to pay any other amount which such Loan Party is obligated to pay under this Agreement or any other Loan Document or which may be required to protect or preserve the Collateral, Bank may obtain such insurance or make such payment, and all amounts so paid by Bank are Bank Expenses and immediately due and payable, bearing interest at the then highest rate applicable to the Obligations, and secured by the Collateral. Bank will make reasonable efforts to provide such Loan Party with notice of Bank obtaining such insurance at the time it is obtained or within a reasonable time thereafter. No payments by Bank are deemed an agreement to make similar payments in the future or Bank's waiver of any Event of Default.
- 9.4 Application of Payments and Proceeds Upon Default. If an Event of Default has occurred and is continuing, Bank shall have the right to apply in any order any funds in its possession and constituting Collateral (other than Excluded Accounts), whether from such Loan Party's account balances, payments, proceeds realized as the result of any collection of Accounts or other disposition of the Collateral, or otherwise, to the Obligations. Bank shall pay any surplus to the Loan Parties by credit to the Designated Deposit Account or to other Persons legally entitled thereto; the Loan Parties shall remain liable to Bank for any deficiency. If Bank, directly or indirectly, enters into a deferred payment or other credit transaction with any purchaser at any sale of Collateral, Bank shall have the option, exercisable at any time, of either reducing the Obligations by the principal amount of the purchase price or deferring the reduction of the Obligations until the actual receipt by Bank of cash therefor.
- 9.5 <u>Bank's Liability for Collateral</u>. So long as Bank complies with reasonable banking practices regarding the safekeeping of the Collateral in the possession or under the control of Bank, Bank shall not be liable or responsible for: (a) the safekeeping of the Collateral; (b) any loss or damage to the Collateral; (c) any diminution in the value of the Collateral; or (d) any act or default of any carrier, warehouseman, bailee, or other Person. The Loan Parties bear all risk of loss, damage or destruction of the Collateral.
- 9.6 No Waiver; Remedies Cumulative. Bank's failure, at any time or times, to require strict performance by the Loan Parties of any provision of this Agreement or any other Loan Document or any Bank Services Agreement shall not waive, affect, or diminish any right of Bank thereafter to demand strict performance and compliance herewith or therewith. No waiver hereunder shall be effective unless signed by the party granting the waiver and then is only effective for the specific instance and purpose for which it is given. Bank's rights and remedies under this Agreement, the other Loan Documents and any Bank Services Agreement are cumulative. Bank has all rights and remedies provided under the Code, by law, or in equity. Bank's exercise of one right or remedy is not an election and shall not preclude Bank from exercising any other remedy under this Agreement or other remedy available at law or in equity, and Bank's waiver of any Event of Default is not a continuing waiver. Bank's delay in exercising any remedy is not a waiver, election, or acquiescence.

- 42. -

9.7 <u>Demand Waiver</u>. Each Loan Party waives demand, notice of default or dishonor, notice of payment and nonpayment, notice of any default, nonpayment at maturity, release, compromise, settlement, extension, or renewal of accounts, documents, instruments, chattel paper, and guarantees held by Bank on which such Loan Party is liable.

Section 10 Notices.

All notices, consents, requests, approvals, demands, or other communication by any party to this Agreement or any other Loan Document must be in writing and shall be deemed to have been validly served, given, or delivered: (a) upon the earlier of actual receipt and three (3) Business Days after deposit in the U.S. mail, first class, registered or certified mail return receipt requested, with proper postage prepaid; (b) upon transmission, when sent by electronic mail; (c) one (1) Business Day after deposit with a reputable overnight courier with all charges prepaid; or (d) when delivered, if hand-delivered by messenger, all of which shall be addressed to the party to be notified and sent to the address or email address indicated below. Bank or any Loan Party may change its mailing or electronic mail address by giving the other party written notice thereof in accordance with the terms of this Section 10.

If to any Loan Party:

Koru Medical Systems, Inc. 100 Corporate Drive Mahwah, New Jersey 07430 Attn: Thomas Adams

Email: Tadams@korumedical.com

With a copy to:

Two Logan Square

100 N. 18th Street, Suite 710 Philadelphia, Pennsylvania 19103 Attn: Heather R. Badami, Esq. Email: hbadami@rccblaw.com

If to Bank: HSBC Ventures USA Inc.

c/o CRE Servicing 2929 Walden Ave. Depew, NY 14043 Phone: 716-841-6444

Email: creservicing@us.hsbc.com

With a copy to: HSBC Ventures USA Inc.

Attention: Loan Operations 545 Washington Blvd, 9th Floor

Jersey City, NJ 07310

Email: creservicing@us.hsbc.com

and: Morrison & Foerster LLP

200 Clarendon Street, Floor 21 Boston, Massachusetts 02116 Attn: David A. Ephraim, Esquire Email: DEphraim@mofo.com

Section 11 Choice of Law, Venue, and Jury Trial Waiver.

- 43 -

This Agreement and the other Loan Documents and any claim, controversy, dispute or cause of action (whether in contract, tort or otherwise) arising out of or relating thereto (except, as to any Loan Document, as expressly set forth therein) and the transactions contemplated by such documents shall be governed by, and construed in accordance with, the law of the State of New York, without regard to conflicts of law principles except Title 14 of Article 5 of the New York General Obligations law. Each of the parties hereto hereby irrevocably consents to the non-exclusive jurisdiction of the courts of the State of New York and of any federal court located in the Borough of Manhattan in such State in connection with any action, suit or other proceeding arising out of or relating to this Agreement or any action taken or omitted hereunder, and waives any claim of forum non conveniens and any objections as to laying of venue. Nothing in this Agreement shall affect any right Bank may otherwise have to bring any action or proceeding relating to this Agreement or the other Loan Documents against any of the Loan Parties or its properties in the courts of any jurisdiction. Each party further waives personal service of any summons, complaint or other process, right to a jury trial and agrees that service thereof may be made by certified or registered mail directed to such person at such person's address for purposes of notices hereunder and that service so made shall be deemed completed upon the earlier to occur of such person's actual receipt thereof or three (3) days after deposit in the U.S. mails, proper postage prepaid.

TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH LOAN PARTY AND BANK EACH WAIVE THEIR RIGHT TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION ARISING OUT OF OR BASED UPON THIS AGREEMENT, THE LOAN DOCUMENTS OR ANY CONTEMPLATED TRANSACTION, INCLUDING CONTRACT, TORT, BREACH OF DUTY AND ALL OTHER CLAIMS. THIS WAIVER IS A MATERIAL INDUCEMENT FOR BOTH PARTIES TO ENTER INTO THIS AGREEMENT. EACH PARTY HAS REVIEWED THIS WAIVER WITH ITS COUNSEL.

This Section 11 shall survive the termination of this Agreement.

Section 12 General Provisions.

12.1 <u>Termination Prior to Maturity Date; Survival</u>. All covenants, representations and warranties made in this Agreement shall continue in full force until this Agreement has terminated pursuant to its terms and all Obligations (other than inchoate indemnity obligations, any other obligations which, by their terms, are to survive the termination of this Agreement, and any Obligations under Bank Services Agreements that are cash collateralized in accordance with Section 4.1 of this Agreement) have been satisfied. So long as the Loan Parties have satisfied the Obligations (other than inchoate indemnity obligations, any other obligations which, by their terms, are to survive the termination of this Agreement, and any Obligations under Bank Services Agreements that are cash collateralized in accordance with Section 4.1 of this Agreement), this Agreement may be terminated prior to the Revolving Line Maturity Date and Term Loan Maturity Date by Borrower, effective three (3) Business Days after written notice of termination is given to Bank. Those obligations that are expressly specified in this Agreement as surviving this Agreement's termination shall continue to survive notwithstanding this Agreement's termination.

Successors and Assigns. This Agreement binds and is for the benefit of the successors and permitted assigns of each party. No Loan Party may assign this Agreement or any rights or obligations under it without Bank's prior written consent (which may be granted or withheld in Bank's discretion). Bank has the right, without the consent of any Loan Party, to sell, transfer, assign, negotiate, or grant participation in all or any part of, or any interest in, Bank's obligations, rights, and benefits under this Agreement and the other Loan Documents. So long as no Event of Default is continuing, Bank shall provide Borrower with five (5) Business Days' prior notice of any such assignment; provided that (a) the failure to provide such notice shall not affect the validity of such assignment or result in any liability of Bank and (b) no such notice shall be required in connection with any participation. Notwithstanding the

- 44 -

above, Bank may at any time without notice to Borrower pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of Bank (or its Affiliates) including without limitation, any pledge or assignment to secure obligations to a Federal Reserve Bank in accordance with applicable law. Bank, acting solely for this purpose as an agent of Borrower, shall maintain a copy of each assignment or other transfer delivered to it and a register for the recordation of the names and addresses of the assignee(s), transferee(s), participant(s) or other recipient(s), and the commitments of, and principal amounts (and stated interest) of the Credit Extensions owing to, each such assignee, transferee, participant or other recipient pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive absent manifest error, and Borrower and Bank shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a lender hereunder for all purposes of this Agreement. The Register shall be available for inspection by Borrower, Bank and any assignee, at any reasonable time and from time to time upon reasonable prior notice. This Section 12.2 shall be construed so that the Credit Extensions are at all times maintained in "registered form" within the meaning of Section 5f.103-1(c) or proposed Section 1.163-5(b) of the U.S. Treasury Regulations promulgated under the IRC. Each lender that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of Borrower, maintain a register on which it enters the name and address of each participant and the principal amounts (and stated interest) of each participant's interest in the loans or other obligations under the Loan Documents (the "Participant Register"); provided that no such lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any participant or any information relating to a participant's interest in any commitments, loans, letters of credit or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan, letter of credit or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary.

12.3 <u>Indemnification</u>. Each Loan Party agrees to indemnify, defend and hold Bank and its directors, officers, employees, agents, attorneys, or any other Person affiliated with or representing Bank (each, an "Indemnified Person") harmless against: (a) all obligations, demands, claims, and liabilities (collectively, "Claims") claimed or asserted by any other party under or in connection with the transactions contemplated by the Loan Documents and any Bank Services Agreement; and (b) all losses or expenses (including Bank Expenses) in any way suffered, incurred, or paid by such Indemnified Person as a result of, following from, or arising from transactions between Bank and any Loan Party (including reasonable attorneys' fees and expenses) under or in connection with the Loan Documents and any Bank Services Agreement, except for Claims and/or losses directly (x) caused by any Indemnified Person's gross negligence or willful misconduct as determined by a final and non-appealable decision of a court of competent jurisdiction or (y) resulting from a claim brought by a Loan Party against an Indemnified Person for any material breach of such Indemnified Person's obligations hereunder or under another Loan Document, if such Loan Party has obtained a final and non-appealable judgment in its favor on such claim as determined by a court of competent jurisdiction. This Section 12.3 shall not apply with respect to Taxes other than any Taxes that represent losses or expenses arising from any non-Tax claim.

This Section 12.3 shall survive until all statutes of limitation with respect to the Claims, losses, and expenses for which indemnity is given shall have run.

- 12.4 <u>Time of Essence</u>. Time is of the essence for the performance of all Obligations in this Agreement.
- 12.5 <u>Severability of Provisions</u>. Each provision of this Agreement is severable from every other provision in determining the enforceability of any provision.

- 45 -

- 12.6 <u>Correction of Loan Documents</u>. Bank may correct patent errors and fill in any blanks in the Loan Documents consistent with the agreement of the parties.
- 12.7 <u>Amendments in Writing; Waiver; Integration</u>. No purported amendment or modification of any Loan Document, or waiver, discharge or termination of any obligation under any Loan Document, shall be enforceable or admissible unless, and only to the extent, expressly set forth in a

writing signed by the party against which enforcement or admission is sought. Without limiting the generality of the foregoing, no oral promise or statement, nor any action, inaction, delay, failure to require performance or course of conduct shall operate as, or evidence, an amendment, supplement or waiver or have any other effect on any Loan Document. Any waiver granted shall be limited to the specific circumstance expressly described in it, and shall not apply to any subsequent or other circumstance, whether similar or dissimilar, or give rise to, or evidence, any obligation or commitment to grant any further waiver. The Loan Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of the Loan Documents merge into the Loan Documents.

- 12.8 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, is an original, and all taken together, constitute one Agreement.
- 12.9 Confidentiality. In handling any confidential information regarding the Loan Parties and their respective Subsidiaries and their business, Bank shall exercise the same degree of care that it exercises for its own proprietary information, but disclosure of information may be made: (a) to Bank's Subsidiaries or Affiliates (such Subsidiaries and Affiliates, together with Bank, collectively, "Bank Entities") (provided that such Bank Entities have agreed to the terms of this provision or to terms substantially the same as those of this provision); (b) to prospective transferees or purchasers of any interest in the Credit Extensions (provided, however, Bank shall obtain any prospective transferee's or purchaser's agreement to the terms of this provision or to terms substantially the same as those of this provision); (c) as required by law, regulation, subpoena, or other order; (d) to Bank's regulators or as otherwise required in connection with Bank's examination or audit; (e) as Bank considers appropriate in exercising remedies under the Loan Documents; and (f) to third-party service providers of Bank so long as such service providers have executed a confidentiality agreement with Bank with terms no less restrictive than those contained herein. Confidential information does not include information that is either: (i) in the public domain or in Bank's possession when disclosed to Bank, or becomes part of the public domain (other than as a result of its disclosure by Bank in violation of this Agreement) after disclosure to Bank; or (ii) disclosed to Bank by a third party, if Bank does not know that the third party is prohibited from disclosing the information. Bank Entities may use confidential information for the development of databases, reporting purposes, and market analysis so long as such confidential information is aggregated and anonymized prior to distribution unless otherwise expressly permitted by Borrower. The provisions of the immediately preceding sentence shall survive the termination of this Agreement.

12.10 Taxes.

(a) Except as required by applicable law, any and all payments by any Loan Party under any Loan Document shall be made without deduction or withholding for any Taxes, except as required by applicable law. If any applicable law requires the deduction or withholding of any Tax from any such payment by any Loan Party, then the applicable Loan Party shall be entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant governmental authority in accordance with applicable law, and, if such Tax is imposed on or with respect

- 46 -

to any payment made by or on account of any obligation of such Loan Party under any Loan Document or Other Taxes (an "Indemnified Tax"), then the sum payable by such Loan Party shall be increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this Section), Bank receives an amount equal to the sum it would have received had no such deduction or withholding been made; provided, however, that no participant shall be entitled to receive any greater payment under this Section 12.10(a) with respect to any participation than the participating or assigning Bank would have been entitled to receive, except to the extent such entitlement to receive a greater payment results from a Change in Law that occurs after the participant acquired the applicable participation. For purposes of this Section 12.10, the term "applicable law" includes FATCA. Notwithstanding anything in the Loan Documents, Indemnified Taxes shall not include Excluded Taxes.

- (b) Each Loan Party hereby indemnifies Bank, within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section) payable or paid by Bank or required to be withheld or deducted from a payment to Bank and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant governmental authority. Promptly upon having knowledge that any such Indemnified Taxes have been levied, imposed or assessed, and promptly upon notice by Bank, such Loan Party shall pay such Indemnified Taxes directly to the relevant taxing authority or governmental authority; provided that Bank shall not be under any obligation to provide any such notice to any Loan Party. A certificate as to the amount of such payment or liability delivered to Borrower by Bank shall be conclusive absent manifest error.
- (c) If Bank is entitled to an exemption from or reduction of withholding tax with respect to payments made under any Loan Document, at Borrower's request Bank shall deliver to

Borrower at the time or times reasonably requested by Borrower, such properly completed and executed documentation reasonably requested by Borrower as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, Bank, if reasonably requested by Borrower, shall deliver such other documentation prescribed by applicable law or reasonably requested by Borrower as will enable Borrower to determine whether or not Bank is subject to backup withholding or information reporting requirements. Each Person that becomes a successor, assignee or participant (or other transferee) of Bank pursuant to Section 12.2 shall, upon the effectiveness of the related transfer, be required to provide all the forms and statements required pursuant to this Section 12.10(c), as if it were "Bank" (it being understood that for participants the documentation required under this Section shall be delivered to the participating Bank).

- (d) As soon as practicable after any payment of Taxes by any Loan Party to a Governmental Authority pursuant to this Section, the applicable Loan Party shall deliver to Bank the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to Bank
- (e) Each party's obligations under this Section shall survive any assignment of rights by, or the replacement of, Bank, the termination of the Credit Extensions and the repayment, satisfaction or discharge of all obligations under any Loan Document.
- 12.11 <u>Attorneys' Fees, Costs and Expenses</u>. In any action or proceeding between any Loan Party and Bank arising out of or relating to the Loan Documents, Bank shall be entitled to recover its reasonable attorneys' fees and other costs and expenses incurred, in addition to any other relief to

- 47 -

which it may be entitled, including without limitation, in connection with any appeals, and the enforcement of any judgment or similar order.

- 12.12 <u>Waiver of Consequential Damages, Etc.</u> To the fullest extent permitted by Applicable Law, the Borrower shall not assert, and hereby waives, any claim against Bank, or any Affiliate of Bank, any of their respective officers, directors, managers, employees, agents, attorneys or representatives, of any of the foregoing Persons (each such Person being called a "Protected Person"), on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any loan or letter of credit, or the use of the proceeds thereof. No Protected Person shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby.
- 12.13 <u>Electronic Execution of Documents</u>. The words "execution," "signed," "signature" and words of like import in any Loan Document shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity and enforceability as a manually executed signature or the use of a paper-based recordkeeping systems, as the case may be, to the extent and as provided for in any applicable law, including, without limitation, any state law based on the Uniform Electronic Transactions Act.
- 12.14 <u>Right of Setoff</u>. Each Loan Party hereby grants to Bank a Lien and a right of setoff as security for all Obligations to Bank, whether now existing or hereafter arising upon and against all deposits, credits, collateral and property, now or hereafter in the possession, custody, safekeeping or control of Bank or any entity under the control of Bank (including a subsidiary of Bank) or in transit to any of them. At any time after the occurrence and during the continuance of an Event of Default, without demand or notice, Bank may setoff the same or any part thereof (other than amounts in deposit accounts exclusively used for payroll, payroll taxes, and other employee wage and benefit payments to or for the benefit of such Loan Party's employees and identified to Bank by Borrower as such) and apply the same to any liability or Obligation of such Loan Party even though unmatured and regardless of the adequacy of any other collateral securing the Obligations. ANY AND ALL RIGHTS TO REQUIRE BANK TO EXERCISE ITS RIGHTS OR REMEDIES WITH RESPECT TO ANY OTHER COLLATERAL WHICH SECURES THE OBLIGATIONS, PRIOR TO EXERCISING ITS RIGHT OF SETOFF WITH RESPECT TO SUCH DEPOSITS, CREDITS OR OTHER PROPERTY OF ANY LOAN PARTY, ARE HEREBY KNOWINGLY, VOLUNTARILY AND IRREVOCABLY WAIVED.
- 12.15 <u>Captions</u>. The headings used in this Agreement are for convenience only and shall not affect the interpretation of this Agreement.
- 12.16 <u>Construction of Agreement</u>. The parties mutually acknowledge that they and their attorneys have participated in the preparation and negotiation of this Agreement. In cases of uncertainty this Agreement shall be construed without regard to which of the parties caused the uncertainty to exist.
- 12.17 <u>Relationship.</u> The relationship of the parties to this Agreement is determined solely by the provisions of this Agreement. The parties do not intend to create any agency, partnership,

- 12.18 <u>Third Parties</u>. Nothing in this Agreement, whether express or implied, is intended to: (a) confer any benefits, rights or remedies under or by reason of this Agreement on any persons other than the express parties to it and their respective permitted successors and assigns; (b) relieve or discharge the obligation or liability of any person not an express party to this Agreement; or (c) give any person not an express party to this Agreement any right of subrogation or action against any party to this Agreement.
- 12.19 <u>Patriot Act; Compliance with Sanctions</u>. Pursuant to the requirements of the Patriot Act, Borrower must provide information to Bank that enables it to verify identification information concerning Borrower, including the name and address of Borrower and other information that will allow Bank to identify Borrower in accordance with the Patriot Act.
- 12.20 <u>Marketing Consent</u>. Borrower hereby authorizes Bank, at its sole expense, but without any prior approval by Borrower, to publish such tombstones and give such other publicity to this Agreement as it may from time to time determine in its sole discretion. The foregoing authorization shall remain in effect unless and until Borrower notifies Bank in writing that such authorization is revoked.

[Signature page follows.]

- 49 -

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed under seal as of the Effective Date.

BORROWER:

KORU MEDICAL SYSTEMS, INC.

By /s/Thomas Adams

Name: Thomas Adams

Title: Chief Financial Officer

BANK:

HSBC VENTURES USA INC.

By /s/ Evan Chen

Name: Evan Chen Title: Vice President

- 50 -

EXHIBIT A

COLLATERAL DESCRIPTION

The Collateral consists of all of Borrower's right, title and interest in and to the following personal property:

(a) (i) all goods, Accounts (including health-care receivables), Equipment, Inventory, contract rights or rights to payment of money, leases, license agreements, franchise agreements, General Intangibles (except as provided below), commercial tort claims, documents, instruments (including any promissory notes), chattel paper (whether tangible or electronic), cash, deposit accounts, certificates of deposit, fixtures, letters of credit rights (whether or not the letter of credit is evidenced by a writing), securities, securities accounts, securities entitlements and all other investment property, supporting obligations, and financial assets, whether now owned or hereafter acquired, wherever located; and (ii) all Books relating to the foregoing, and any and all claims, rights and interests in any of the above and all substitutions for, additions, attachments, accessories, accessions and improvements to and replacements, products, proceeds and insurance proceeds of any or all of the foregoing.

(b) Notwithstanding the foregoing, the Collateral does not include any Intellectual Property; provided, however, the Collateral shall include all Accounts and all proceeds of Intellectual Property. If a judicial authority (including a U.S. Bankruptcy Court) would hold that a security interest in the underlying Intellectual Property is necessary to have a security interest in such Accounts and such property that are proceeds of Intellectual Property, then the Collateral shall automatically, and effective as of the Effective Date, include the Intellectual Property to the extent necessary to permit perfection of Bank's security interest in such Accounts and such other property of Borrower that are proceeds of the Intellectual Property.

Exhibit 10.2

THIS WARRANT AND THE SHARES ISSUABLE HEREUNDER HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AS SET FORTH IN SECTIONS 6.3 AND 6.4 BELOW, MAY NOT BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND LAWS OR, IN THE OPINION OF LEGAL COUNSEL IN FORM AND SUBSTANCE SATISFACTORY TO THE COMPANY, SUCH OFFER, SALE, PLEDGE OR OTHER TRANSFER IS EXEMPT FROM SUCH REGISTRATION.

STOCK PURCHASE WARRANT

Company: KORU Medical Systems, Inc., a Delaware corporation

Number of Shares: As set forth in Section 1 below

Class/Series of Stock: Common Stock, \$0.01 par value per share Exercise Price: \$2.12 per Share, subject to adjustment as provided herein

Issue Date: March 8, 2024 Expiration Date: March 7, 2034

Credit Facility: This Stock Purchase Warrant (as amended and in effect from time to time, this "Warrant") is issued in connection with that certain Loan and Security Agreement of even date herewith among HSBC Ventures USA Inc., the Company and the other parties named therein (as amended and/or

modified and in effect from time to time, the "Loan Agreement").

THIS WARRANT PROVIDES THAT, for good and valuable consideration, HSBC VENTURES USA INC. (together with any successor or permitted assignee or transferee of this Warrant or of any shares issued upon exercise hereof, "Holder") is entitled to purchase from the above-named company (the "Company") up to the number of fully paid and non-assessable shares of the above-stated class/series of stock (the "Class") set forth in Section 1 below, at a purchase price per Share equal to the above-stated exercise price (the "Exercise Price"), subject to the provisions and upon the terms and conditions set forth in this Warrant.

SECTION 1. NUMBER OF SHARES.

- 1.1 <u>Number of Shares.</u> This Warrant shall be exercisable for the Initial Shares, plus the Additional Shares, if any (collectively, and as may be adjusted from time to time in accordance with the provisions of this Warrant, the "Shares").
- 1.2 <u>Initial Shares.</u> As used herein, "**Initial Shares**" means 38,052 shares of the Class, subject to adjustment from time to time in accordance with the provisions of this Warrant.
- 1.3 Additional Shares. Upon the making, if any, of the first Term Loan Advance (as defined in the Loan Agreement) to the Company in any amount, this Warrant automatically shall become exercisable for an additional 38,052 shares of the Class, as such number may be adjusted from time to time in accordance with the provisions of this Warrant (the "Additional Shares"), including, without limitation, adjustments in respect of events occurring prior to the date, if any, on which this Warrant becomes exercisable for such shares as if they constituted "Shares" hereunder for such purpose at all times from the Issue Date.

SECTION 2. EXERCISE.

2.1 Method of Exercise. Holder may exercise this Warrant in whole or in part at any time and from time to time prior to the expiration or earlier termination hereof by delivering to the Company the original of this Warrant together with a duly executed Notice of Exercise in substantially the form attached hereto as Appendix 1 and, unless Holder is exercising this Warrant pursuant to a cashless exercise set forth in Section 2.2, a check, wire transfer of same-day funds (to an account designated by the Company), or other form of payment acceptable to the Company for the aggregate Exercise Price for the Shares being purchased. Notwithstanding any contrary provision herein, if this Warrant was originally executed and/or delivered

electronically, in no event shall Holder be required to surrender or deliver an ink-signed paper copy of this Warrant in connection with its exercise hereof or of any rights hereunder, nor shall Holder be required to surrender or deliver a paper or other physical copy of this Warrant in connection with any exercise hereof.

2.2 <u>Cashless Exercise</u>. On any exercise of this Warrant, in lieu of payment of the aggregate Exercise Price in the manner specified in Section 2.1 above (but otherwise in accordance with the requirements of such Section), Holder may elect to surrender to the Company Shares having an aggregate value equal to the aggregate Exercise Price. If Holder makes such election, the Company shall issue to Holder such number of fully paid and non-assessable Shares determined by the following formula:

$$X = Y(A-B)/A$$

where:

- Y = the number of Shares with respect to which this Warrant is being exercised (inclusive of the Shares surrendered to the Company in payment of the aggregate Exercise Price);
- A = the then-fair market value (as determined pursuant to Section 2.3 below) of one Share; and
- B = the then-effective Exercise Price.
- 2.3 <u>Fair Market Value</u>. If shares of the Class are then traded or quoted on a nationally recognized securities exchange, inter-dealer quotation system or over-the-counter market (a "<u>Trading Market</u>"), the fair market value of a Share shall be the closing price or last sale price of a share of the Class reported for the Business Day immediately before the date on which Holder delivers this Warrant together with its Notice of Exercise to the Company. If shares of the Class are not then traded in a Trading Market, the Board of Directors of the Company shall determine the fair market value of a Share in its reasonable good faith judgment.
- 2.4 <u>Delivery of Certificate and New Warrant</u>. Within a reasonable time after Holder exercises this Warrant in the manner set forth in Section 2.1 or 2.2 above, the Company shall deliver to Holder a certificate, which may be in electronic form (or, in the case of uncertificated securities, provide notice of book entry) representing the Shares issued to Holder upon such exercise and, if this Warrant has not been fully exercised and has not expired, a new warrant of like tenor representing the Shares not so acquired (or surrendered in payment of the aggregate Exercise Price).

2.5 Replacement of Warrant.

- (a) <u>Paper Original Warrant</u>. To the extent that the original of this Warrant is a paper original, on receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant and, in the case of loss, theft or destruction, on delivery of an indemnity agreement reasonably satisfactory in form, substance and amount to the Company or, in the case of mutilation, on surrender of this Warrant to the Company for cancellation, the Company shall, within a reasonable time, execute and deliver to Holder, in lieu of this Warrant, a new warrant of like tenor and amount.
- (b) <u>Electronic Original Warrant</u>. To the extent that the original of this Warrant is an electronic original, if at any time this Warrant is rejected by any person (including, but not limited to, paying or escrow agents) or any such person fails to comply with the terms of this Warrant based on this Warrant being presented to such person as an electronic record or a printout hereof, or any signature hereto being in electronic form, the Company shall, promptly upon Holder's request and without indemnity, execute and deliver to Holder,

- 2 -

in lieu of electronic original versions of this Warrant, a new warrant of like tenor and amount in paper form with original ink signatures.

2.6 Treatment of Warrant Upon Acquisition of Company.

- (a) <u>Acquisition</u>. For the purpose of this Warrant, "<u>Acquisition</u>" means any transaction or series of related transactions involving: (i) the sale, lease, exclusive license, or other disposition of all or substantially all of the assets of the Company; (ii) any merger or consolidation of the Company into or with another person or entity (other than a merger or consolidation effected exclusively to change the Company's domicile), or any other corporate reorganization, in which the stockholders of the Company in their capacity as such immediately prior to such merger, consolidation or reorganization, own less than a majority of the Company's (or the surviving or successor entity's) outstanding voting power immediately after such merger, consolidation or reorganization; (iii) any merger, consolidation or other business combination of the Company into or with a special purpose acquisition company or wholly-owned subsidiary thereof; or (iv) any sale or other transfer by the stockholders of the Company of shares representing a majority of the Company's then-total outstanding combined voting power. For the avoidance of doubt, "Acquisition" shall not include any sale and issuance by the Company of shares of its capital stock or of securities or instruments exercisable for or convertible into, or otherwise representing the right to acquire, shares of its capital stock to one or more investors for cash in a transaction or series of related transactions the primary purpose of which is a bona fide equity financing of the Company.
- (b) Treatment of Warrant in Cash/Public Company Acquisition. In the event of an Acquisition in which the consideration to be received by the holders of the outstanding shares of the Class (in their capacity as such) consists solely of cash, solely of Marketable Securities (as hereinafter defined) or a combination of cash and Marketable Securities (a "Cash/Public Company Acquisition"), and the fair market value of one Share as determined in accordance with Section 2.3 above would be greater than the Exercise Price in effect as of immediately prior to the closing of such Cash/Public Company Acquisition, and Holder has not previously exercised this Warrant in full, then, in lieu of Holder's exercise of the unexercised portion of this Warrant, this Warrant shall, as of immediately prior to such closing (but subject to the occurrence thereof), automatically cease to represent the right to purchase Shares and shall, from and after such closing, represent solely the right to receive the aggregate consideration that would have been payable in such Cash/Public Company Acquisition on and in respect of all Shares for which this Warrant was exercisable as of immediately prior to the closing thereof, net of the aggregate Exercise Price therefor, as if such Shares had been issued and

outstanding to Holder as of immediately prior to such closing, as and when such consideration is paid to the holders of the outstanding shares of the Class. In the event of a Cash/Public Company Acquisition in which the fair market value of one Share as determined in accordance with Section 2.3 above would be equal to or less than the Exercise Price in effect as of immediately prior to the closing of such Cash/Public Company Acquisition, then this Warrant will automatically and without further action of any party terminate as of immediately prior to such closing

- (c) Treatment of Warrant in non-Cash/Public Company Acquisition. Upon the closing of any Acquisition other than a Cash/Public Company Acquisition, the acquiring, surviving or successor entity shall assume this Warrant and the Company's obligations hereunder, and this Warrant shall thereafter be exercisable for the same securities and/or other property as would have been paid for the Shares issuable upon exercise of the unexercised portion of this Warrant as if such Shares were outstanding on and as of the closing of such Acquisition, at an aggregate Exercise Price equal to the aggregate Exercise Price in effect as of immediately prior to such closing, all subject to further adjustment from time to time thereafter in accordance with the provisions of this Warrant.
- (d) As used in this Warrant, "Marketable Securities" means securities meeting all of the following requirements (determined as of immediately prior to the closing of the Acquisition): (i) the issuer thereof is then subject to the reporting requirements of Section 13 or Section 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and is then current in its filing of all required reports and other information under the Act and the Exchange Act; (ii) the class and series of shares or other security of the issuer

- 3 -

that would be received by Holder in connection with the Acquisition were Holder to exercise this Warrant on or prior to the closing thereof is then traded in a Trading Market; and (iii) following the closing of such Acquisition, Holder would not be restricted from publicly re-selling all of the issuer's shares and/or other securities that would be received by Holder in such Acquisition were Holder to exercise this Warrant in full on or prior to the closing of such Acquisition, except to the extent that any such restriction (x) arises solely under federal or state securities laws, rules or regulations, and (y) does not extend beyond six (6) months from the closing of such Acquisition. Notwithstanding the foregoing provisions of this Section 2.6(d), securities held in escrow or subject to holdback to cover indemnification-related claims shall be deemed to be Marketable Securities if they would otherwise be Marketable Securities but for the fact that they are held in escrow or subject to holdback to cover indemnification-related claims.

SECTION 3. CERTAIN ADJUSTMENTS TO THE SHARES, CLASS AND EXERCISE PRICE.

- 3.1 <u>Stock Dividends, Splits, Etc.</u> If the Company declares or pays a dividend or distribution on the outstanding shares of the Class payable in additional shares of the Class or other securities or property (other than cash), then upon exercise of this Warrant, for each Share acquired, Holder shall receive, without additional cost to Holder, the total number and kind of securities and property which Holder would have received had Holder owned the Shares of record as of the date the dividend or distribution occurred. If the Company subdivides the outstanding shares of the Class by reclassification or otherwise into a greater number of shares, the number of Shares purchasable hereunder shall be proportionately increased and the Exercise Price shall be proportionately decreased. If the outstanding shares of the Class are combined or consolidated, by reclassification or otherwise, into a lesser number of shares, the Exercise Price shall be proportionately increased and the number of Shares shall be proportionately decreased.
- 3.2 <u>Reclassification, Exchange, Combinations or Substitution</u>. Upon any event whereby all of the outstanding shares of the Class are reclassified, exchanged, combined, substituted, or replaced for, into, with or by Company securities of a different class and/or series, then from and after the consummation of such event, "Class" shall mean such securities and this Warrant will be exercisable for the number of such securities that Holder would have received had the Shares been outstanding on and as of the consummation of such event, at an aggregate Exercise Price equal to the aggregate Exercise Price in effect as of immediately prior to such event, all subject to further adjustment thereafter from time to time in accordance with the provisions of this Warrant. The provisions of this Section 3.2 shall similarly apply to successive reclassifications, exchanges, combinations, substitutions, replacements or other similar events.
- 3.3 Adjustment to Exercise Price on Cash Dividend. In the event that the Company at any time or from time to time prior to the exercise in full of this Warrant pays any cash dividend on the outstanding shares of the Class or makes any cash distribution on or in respect of all outstanding shares of the Class (other than a distribution of cash proceeds received by the Company in connection with an Acquisition described in Section 2.6(a)(i) above), then on and as of the date of each such dividend payment and/or distribution, the Exercise Price shall be reduced by an amount equal to the amount paid or distributed upon or in respect of each outstanding share of the Class; provided that in no event shall the Exercise Price be reduced below the then-par value, if any, of a share of the Class.
- 3.4 <u>No Fractional Share</u>. No fractional Share shall be issued upon exercise of this Warrant, and the number of Shares to be issued shall be rounded down to the nearest whole Share. If a fractional Share interest arises upon any exercise of this Warrant, the Company shall eliminate such fractional Share interest by paying Holder in cash an amount equal to (a) such fractional interest, multiplied by (b)(i) the fair market value (as determined in accordance with Section 2.3 above) of a full Share, less (ii) the then-effective Exercise Price.

3.5 <u>Certificate as to Adjustments</u>. Within a reasonable time following each adjustment of the Exercise Price, Class and/or number of Shares pursuant to the terms of this Warrant, the Company at its expense shall deliver a certificate of its Chief Financial Officer or other authorized officer to Holder setting forth the adjustments to the Exercise Price, Class and/or number of Shares and the facts upon which such adjustments are based. The Company shall, at any time and from time to time within a reasonable time following Holder's written

- 4 -

request and at the Company's expense, furnish Holder with a certificate of its Chief Financial Officer or other authorized officer setting forth the then-current Exercise Price, Class and number of Shares and the computations or other determinations thereof.

SECTION 4. REPRESENTATIONS AND COVENANTS OF THE COMPANY.

4.1 <u>Representations and Warranties</u>. The Company represents and warrants to, and agrees with, Holder as follows:

(a) [Reserved]

- (b) The number of Initial Shares first set forth above together with the number of Additional Shares first set forth above collectively represent not less than 0.150% of the Company's total issued and outstanding shares of capital stock, calculated on and as of the Issue Date hereof on a fully-diluted, common stock-equivalent basis (but without excluding shares of capital stock that are not convertible into shares of common stock) assuming (i) the conversion into common stock of all outstanding securities and instruments (including, without limitation, securities deemed to be outstanding pursuant to clause (ii) of this Section 4.1(b)) convertible by their terms into shares of common stock (regardless of whether such securities or instruments are by their terms now so convertible); (ii) the exercise in full of all outstanding options, warrants (including, without limitation, this Warrant) and other rights to purchase or acquire shares of common stock or securities exercisable for or convertible into shares of common stock (regardless of whether such options, warrants or other rights to purchase or acquire are by their terms now exercisable); and (iii) the inclusion of all shares of common stock reserved for issuance under all of the Company's incentive stock and stock option plans and not now subject to outstanding grants or options.
- (c) All Shares which may be issued upon the exercise of this Warrant, shall, upon issuance, be duly authorized, validly issued, fully paid and non-assessable, and free of any liens and encumbrances except for restrictions on transfer provided for herein or under applicable federal and state securities laws. The Company covenants that it shall at all times cause to be reserved and kept available out of its authorized and unissued capital stock such number of shares of the Class and other securities as will be sufficient to permit the exercise in full of this Warrant.
- (d) The Company's capitalization table attached hereto as <u>Schedule 1</u> is true and complete, in all material respects, as of the Issue Date.
 - 4.2 <u>Notice of Certain Events</u>. If the Company proposes at any time to:
- (a) declare any dividend or distribution upon the outstanding shares of the Class, whether in cash, stock or other securities or property and whether or not a regular cash dividend;
- (b) offer for subscription or sale pro rata to all holders of the outstanding shares of the Class any additional securities of the Company (other than pursuant to contractual pre-emptive or first refusal rights);
- (c) effect any reclassification, exchange, combination, substitution, reorganization or recapitalization of the outstanding shares of the Class; or
 - (d) effect an Acquisition, or to liquidate, dissolve or wind up the Company; or

then, in connection with each such event, the Company shall give Holder notice thereof at the same time and in the same manner as it notifies holders of the outstanding shares of the Class.

4.3 <u>Certain Company Information</u>. The Company will provide such information requested by Holder from time to time, within a reasonable time following each such request, that is reasonably necessary to

- 5 -

enable Holder to comply with Holder's accounting or reporting requirements. Holder agrees to treat and hold all information provided by the Company pursuant to this Warrant in confidence in accordance with the provisions of Section 12.9 of the Loan Agreement (regardless of whether the Loan Agreement shall then be in effect).

4.4 <u>Put Right</u>. Holder shall have the one-time right (but not the obligation), exercisable in its sole discretion on written notice to the Company (the "**Put Notice**") at any time prior to the earlier to occur of (i) exercise in full of this Warrant, and (ii) the expiration or termination of this Warrant, to require the Company to repurchase all (but not less than all) of the unexercised portion of this Warrant from Holder (and the Company

hereby agrees to repurchase this Warrant from Holder upon exercise of such right), free and clear of all liens, claims and encumbrances (except such as may arise by or through the Company), for a total aggregate purchase price of One Dollar (\$1.00), such purchase price to be paid by the Company to Holder in cash or by the Company's check at the Put Closing against surrender by Holder to the Company at or prior to the Put Closing of the original of this Warrant (which may be in electronic form), duly endorsed for transfer on the books of the Company or accompanied by duly executed share transfer powers and/or other instruments of assignment or transfer (the "Put Right"). As used herein, "Put Closing" means the closing of the sale and purchase of this Warrant pursuant to Holder's exercise of the Put Right, on such date (the "Put Closing Date") as shall be set forth in Holder's Put Notice, which date shall be not less than five (5) days following the date of such Put Notice. Notwithstanding anything to the contrary herein, in the event that Holder exercises the Put Right, then on and after the Put Closing Date, regardless of whether the Company shall have tendered payment of the purchase price thereat, this Warrant shall be deemed to have been sold, assigned and transferred by Holder to the Company and shall be the Company's sole and exclusive property.

SECTION 5. REPRESENTATIONS AND COVENANTS OF HOLDER.

Holder represents and warrants to, and agrees with, the Company as follows:

5.1 <u>Investment Representations</u>.

- (a) <u>Purchase for Own Account</u>. This Warrant and the Shares to be acquired upon exercise hereof are being acquired for investment for Holder's account, not as a nominee or agent, and not with a view to the public resale or distribution within the meaning of the Act. Holder also represents that it has not been formed for the specific purpose of acquiring this Warrant or the Shares.
- (b) <u>Disclosure of Information</u>. Holder is aware of the Company's business affairs and financial condition and has received or has had full access to all the information it considers necessary or appropriate to make an informed investment decision with respect to the acquisition of this Warrant and its underlying securities. Holder further has had an opportunity to ask questions of and receive answers from the Company regarding the terms and conditions of the offering of this Warrant and its underlying securities and to obtain additional information (to the extent the Company possessed such information or could acquire it without unreasonable effort or expense) necessary to verify any information furnished to Holder or to which Holder has access.
- (c) <u>Investment Experience</u>. Holder understands that the purchase of this Warrant and its underlying securities involves substantial risk. Holder has experience as an investor in securities of companies in the development stage and acknowledges that Holder can bear the economic risk of such Holder's investment in this Warrant and its underlying securities for an indefinite period of time, and has such knowledge and experience in financial or business matters that Holder is capable of evaluating the merits and risks of its investment in this Warrant and its underlying securities and/or has a preexisting personal or business relationship with the Company and certain of its officers, directors or controlling persons of a nature and duration that enables Holder to be aware of the character, business acumen and financial circumstances of such persons.
- (d) <u>Accredited Investor Status</u>. Holder is an "accredited investor" within the meaning of Regulation D promulgated under the Act.

- 6 -

- (e) The Act. Holder understands that this Warrant and the Shares issuable upon exercise hereof have not been registered under the Act or registered or qualified under the securities laws of any state in reliance upon specific exemptions therefrom, which exemptions depend upon, among other things, the bona fide nature of the Holder's investment intent as expressed herein. Holder understands that the Company is under no obligation to so register or qualify this Warrant, the Shares or such other securities. Holder understands that this Warrant and the Shares issued upon any exercise hereof are "restricted securities" under applicable federal and state securities laws and must be held indefinitely unless subsequently registered under the Act and registered or qualified under applicable state securities laws, or unless exemption from such registration and qualification are otherwise available. Holder is aware of the provisions of Rule 144 promulgated under the Act.
- 5.2 <u>No Stockholder Rights.</u> Without limiting any provision of this Warrant, Holder agrees that as a Holder of this Warrant it will not have any rights (including, but not limited to, voting rights) as a stockholder of the Company with respect to the Shares issuable hereunder unless and until the exercise of this Warrant and then only with respect to the Shares issued on such exercise.

SECTION 6. MISCELLANEOUS.

6.1 <u>Term; Automatic Cashless Exercise Upon Expiration</u>.

- (a) <u>Term.</u> Subject to the provisions of Section 2.6 above, this Warrant is exercisable in whole or in part at any time and from time to time on or before 5:00 PM, Eastern time, on the Expiration Date and shall be void thereafter.
- (b) <u>Automatic Cashless Exercise upon Expiration</u>. In the event that, upon the Expiration Date, the fair market value of one Share as determined in accordance with Section 2.3 above is greater than the Exercise Price in effect on such date, then this Warrant shall automatically be deemed on and as of such

date to have been exercised pursuant to Section 2.2 above as to all Shares for which it shall not previously have been exercised, and the Company shall, within a reasonable time following Holder's written request, deliver a certificate, which may be in electronic form (or, in the case of uncertificated securities, provide notice of book entry) representing the Shares issued to Holder upon such exercise.

6.2 <u>Legends</u>. Each certificate or notice of book entry evidencing Shares shall be imprinted with a legend in substantially the following form:

THE SHARES EVIDENCED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AS SET FORTH IN THAT CERTAIN STOCK PURCHASE WARRANT DATED MARCH 8, 2024 ISSUED BY THE ISSUER TO HSBC VENTURES USA INC., MAY NOT BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND LAWS OR, IN THE OPINION OF LEGAL COUNSEL IN FORM AND SUBSTANCE SATISFACTORY TO THE ISSUER, SUCH OFFER, SALE, PLEDGE OR OTHER TRANSFER IS EXEMPT FROM SUCH REGISTRATION.

6.3 <u>Compliance with Securities Laws on Transfer.</u> This Warrant and the Shares issued upon exercise hereof may not be transferred or assigned in whole or in part except in compliance with applicable federal and state securities laws by the transferor and the transferee (including, without limitation, the delivery of investment representation letters and legal opinions reasonably satisfactory to the Company, as reasonably requested by the Company). The Company shall not require Holder to provide an opinion of counsel in connection with a transfer by Holder of this Warrant if the transfer is to an affiliate (as defined in Regulation D promulgated under the Act) of Holder, provided that such affiliate is an "accredited investor" (as defined in Regulation D promulgated under the Act).

- 7 -

- 6.4 Transfer Procedure. Subject to the provisions of Section 6.3 and upon providing the Company with written notice, Holder may transfer all or part of this Warrant or the Shares issued upon exercise of this Warrant to any transferee; provided that in connection with any such transfer, Holder will give the Company notice of the portion of the Warrant and/or Shares being transferred with the name, address and taxpayer identification number of the transferee, and Holder will surrender this Warrant, or the certificates or other evidence of such Shares or other securities, to the Company for reissuance to the transferee(s) (and to Holder if applicable); and provided further, that any transferee shall make substantially the representations set forth in Section 5.1 above and shall agree in writing with the Company to be bound by all of the terms and conditions of this Warrant.
- Motices. All notices and other communications hereunder from the Company to the Holder, or vice versa, shall be deemed delivered and effective (i) when given personally, (ii) on the third (3rd) Business Day after being mailed by first-class registered or certified mail, postage prepaid, (iii) upon actual receipt if given by electronic mail and such receipt is confirmed in writing electronically by the recipient, or (iv) on the first Business Day following delivery to a reliable overnight courier service, courier fee prepaid, in any case at such address as may have been furnished to the Company or Holder, as the case may be, in writing by the Company or such Holder from time to time in accordance with the provisions of this Section 6.5. All notices to Holder shall be addressed as follows until the Company receives notice of a change of address in connection with a transfer or otherwise:

HSBC Ventures USA Inc. c/o CRE Servicing Larkin U Building 239 Van Rensselaer Street Buffalo, NY 14210 Email: creseervicing@us.hsbc.com

With copies to:

HSBC Ventures USA Inc. Attention: Loan Operations 545 Washington Blvd, 9th Floor Jersey City, NJ 07310 Email: creservicing@us.hsbc.com

HSBC Bank USA, N.A. Attention: Security Services – Operations Manager 452 Fifth Avenue New York, NY 10018

All notices to the Company shall be addressed as follows until Holder receives notice of a change in address:

Koru Medical Systems, Inc. Attn: Chief Financial Officer 100 Corporate Drive Mahwah, NJ 07430

Telephone: (845) 469-

Email: tadams@korumedical.com

With a copy (which shall not constitute notice) to:

Royer Cooper Cohen Braunfeld LLC Attn: Heather R. Badami, Esq.

Two Logan Square 100 N. 18th Street, Suite 710 Philadelphia, PA 19103 Telephone: (215) 839-1000 Email: hbadami@rccblaw.com

- Amendment and Waiver. This Warrant may be amended and any provision hereof waived (either generally or in a particular instance and either retroactively or prospectively) only by an instrument in writing signed by Holder and the Company.
- Attorneys' Fees. In the event of any dispute between Holder and the Company arising out of, based upon or related to this Warrant, the party prevailing in such dispute shall be entitled to collect from the other party all reasonable, documented, out-of-pocket costs and expenses incurred in connection such dispute, including, but not limited to, reasonable attorneys' fees and disbursements.
- Counterparts; Electronic Signatures; Status as Certificated Security. This Warrant may be 68 executed by one or more of the parties hereto in any number of separate counterparts, all of which together shall constitute one and the same instrument. The Company, Holder and any other party hereto may execute this Warrant by electronic means and each party hereto recognizes and accepts the use of electronic signatures and the keeping of records in electronic form by any other party hereto in connection with the execution and storage hereof. To the extent that this Warrant or any agreement subject to the terms hereof or any amendment hereto is executed, recorded or delivered electronically, it shall be binding to the same extent as though it had been executed on paper with an original ink signature, as provided under applicable law, including, without limitation, any state law based on the Uniform Electronic Transactions Act. The fact that this Warrant is executed, signed, stored or delivered electronically shall not prevent the transfer by any Holder of this Warrant pursuant to Section 6.4 or the enforcement of the terms hereof. To the extent that the original of this Warrant is an electronic original, this Warrant, and any copies hereof, shall NOT be deemed to be a "certificated security" within the meaning of Section 8102(a)(4) of the California Commercial Code. Physical possession of the original of this Warrant or any paper copy thereof shall confer no special status to the bearer thereof.
- Headings. The headings in this Warrant are for purposes of reference only and shall not limit or otherwise affect the meaning of any provision of this Warrant.
- Business Days. "Business Day" is any day that is not a Saturday, Sunday or a day on which banks in New York City are required to be closed.

SECTION 7. GOVERNING LAW, VENUE, JURY TRIAL WAIVER.

- 7.1 Governing Law. This Warrant and the parties' respective rights and obligations hereunder shall be governed by and construed in accordance with (i) the General Corporation Law of the State of Delaware. as to all matters within its scope, and (ii) otherwise, the internal domestic laws of the State of New York, without giving effect to its principles regarding conflicts of law.
- Jurisdiction and Venue. The Company and Holder each submit to the non-exclusive jurisdiction of the State and Federal courts in New York, NY; provided, however, that nothing in this Warrant shall be deemed to operate to preclude Holder from bringing suit or taking other legal action in any other jurisdiction to enforce a judgment or other court order in favor of Holder. The Company expressly submits and consents in advance to such jurisdiction in any action or suit commenced in any such court, and the Company hereby waives any objection that it may have based upon lack of personal jurisdiction, improper venue, or forum non conveniens and hereby consents to the granting of such legal or equitable relief as is deemed appropriate by such court. The Company hereby waives personal service of the summons, complaints, and other process issued in such action or suit and agrees that service of such summons, complaints, and other process may be made in accordance with Section 6.5 of this Warrant.

-9-

WARRANT, INCLUDING CONTRACT, TORT, BREACH OF DUTY AND ALL OTHER CLAIMS. THIS WAIVER IS A MATERIAL INDUCEMENT FOR THE PARTIES' AGREEMENT TO THIS WARRANT. EACH PARTY HAS REVIEWED THIS WAIVER WITH ITS COUNSEL.

7.4 <u>Survival</u>. This Section 7 shall survive the termination of this Warrant.

[Remainder of page left blank intentionally]
[Signature page follows]

- 10 -

IN WITNESS WHEREOF, the parties have caused this Stock Purchase Warrant to be executed by their duly authorized representatives effective as of the Issue Date written above.

"COMPANY"

KORU MEDICAL SYSTEMS, INC.

By: __/s/ Thomas Adams
Name: Thomas Adams
Title: Chief Financial Officer

"HOLDER"

HSBC VENTURES USA INC.

By: /s/ Evan Chen
Name: Evan Chen
Title: Vice President



KORU MEDICAL SYSTEMS, INC. ANNOUNCES 2023 FOURTH QUARTER, FULL YEAR FINANCIAL RESULTS, AND 2024 GUIDANCE

MAHWAH, NJ – March 13, 2024 – KORU Medical Systems, Inc. (NASDAQ: KRMD) ("KORU Medical" or the "Company"), a leading medical technology company focused on development, manufacturing, and commercialization of innovative and patient-centric large volume subcutaneous infusion solutions, today reported financial results for the fourth quarter and full year ended December 31, 2023. The Company also issued guidance for the full year 2024.

Recent Highlights

- Fourth quarter 2023 net revenues of \$7.2 million, a 2% decrease from the prior year; Full year 2023 net revenues of \$28.5 million, a 2% increase over the prior year
- Full year 6.7% Core business growth (U.S. and International) driven by overall SCIg market growth, share gains and expansion in new geographies
- Fourth quarter gross margin of 60.3%, reflecting year-over-year improvement of 470 basis points; Full year gross margin of 58.6%, reflecting year-over-year improvement of 350 basis points
- Positive cash flow of \$0.7 million in the fourth quarter of 2023, ending the year with a cash balance of \$11.5 million
- Entered 3 Novel Therapies collaborations during 2023 and announced 2 new collaborations in the first quarter of 2024
- Recorded a non-cash valuation allowance against deferred tax assets of \$6.0 million. Net loss for the year, which included this allowance, was (\$13.7) million, or (\$0.30) per share. Adjusted EBITDA was (\$6.0) million, or (\$0.13) per share vs. (\$6.1) million or (\$0.14) in the prior year.
- Signed a \$10 million credit facility with HSBC USA to replace an expired \$3.5 million revolver agreement; the credit facility
 will be reserved for strategic growth capital
- Issued full year 2024 net revenues guidance of \$31.2-\$32.2 million, representing 10-13% growth

"I am proud of the KORU team for delivering a solid end to 2023, led by performance in our Core business. Entry into multiple geographic markets drove international sales, and in the US, we hit a key milestone with the FDA clearance and launch of the 50mL prefill pump platform, which we anticipate will be a significant growth driver in 2024. Additionally, we saw double-digit growth in our end-user specialty pharmacy sales, reflective of share gains and a growing Ig market" said Linda Tharby, KORU Medical's President and CEO. "Within our Novel Therapies business, we recently announced collaborations for two late-stage therapies. On the operational side, we increased gross margins, were cash flow positive in the fourth quarter and finished the year with a significantly lower cash burn than the prior year. We are excited by the momentum we are carrying into 2024, we remain focused on the execution of our Vision 2026 strategy, and positioning KORU for growth, profitability and value creation for our customers and shareholders."

2023 Fourth Quarter Financial Results

Th	ree Months En	ded D	ecember 31,	C	hange from Pr	ior Year	% of 7	Total
	2023		2022		\$	%	2023	2022
\$	5,565,349	\$	5,314,836	\$	250,513	4.7%	77.4%	72.4%
	1,315,036		1,221,540		93,496	7.7%	18.3%	16.6%
	305,547		808,305		(502,758)	(62.2%)	4.3%	11.0%
\$	7,185,932	\$	7,344,681	\$	(158,749)	(2.2%)		
	\$ \$	\$ 5,565,349 1,315,036 305,547	\$ 5,565,349 \$ 1,315,036 305,547	\$ 5,565,349 \$ 5,314,836 1,315,036 1,221,540 305,547 808,305	2023 2022 \$ 5,565,349 \$ 5,314,836 \$ 1,315,036 \$ 1,221,540 305,547 808,305	2023 2022 \$ 5,565,349 \$ 5,314,836 \$ 250,513 1,315,036 1,221,540 93,496 305,547 808,305 (502,758)	2023 2022 \$ % \$ 5,565,349 \$ 5,314,836 \$ 250,513 4.7% 1,315,036 1,221,540 93,496 7.7% 305,547 808,305 (502,758) (62.2%)	2023 2022 \$ % 2023 \$ 5,565,349 \$ 5,314,836 \$ 250,513 4.7% 77.4% 1,315,036 1,221,540 93,496 7.7% 18.3% 305,547 808,305 (502,758) (62.2%) 4.3%

Total net revenues decreased \$0.2 million, or 2.2%, for the three months ended December 31, 2023, as compared with the same period last year. Domestic Core net revenues increased by 4.7% year-over-year, driven by growth in pumps and consumables, due to competitive share gain in new accounts. International Core net revenues increased by 7.7%, driven by strong SCIg volume and growth in several OUS geographies as we continue to expand into new markets. Novel Therapies net revenues declined by 62% year-over-year, primarily driven by the completion of a large milestone for a pharmaceutical NRE agreement in 2022.

Gross profit increased \$0.3 million or 6.2% for the three months ended December 31, 2023, as compared with the same period last year. Gross profit stated as a percentage of net revenues increased to 60.3% for the fourth quarter of 2023 compared to 55.5% in the fourth quarter of 2022. The increase was primarily driven by production efficiencies from the completion of an outsourced manufacturing initiative and consolidation of our manufacturing sites. Allowing for a one-time inventory adjustment for a product discontinuation, adjusted gross margin for the fourth quarter of 2023 was 63.1%. A reconciliation of adjusted gross margin to the most directly comparable GAAP measure is provided at the end of this press release.

Total operating expenses for the fourth quarter of 2023 were \$6.7 million, an increase of \$0.1 million or 1.6%, compared to \$6.6 million in the prior year period. The increase was driven primarily by the hiring of our new Chief Commercial Officer to lead our Core business growth strategy, partially offset by efficiencies in insourcing R&D.

Net loss for the fourth quarter of 2023 was \$7.5 million, or \$(0.16) per diluted share, compared to a net loss of \$2.0 million, or (\$0.04) per diluted share for the same period of 2022. Net loss included a tax valuation allowance of \$6.0 million for the fourth quarter of 2023. Adjusted EBITDA for the quarter was (\$1.0 million), or (\$0.02) per diluted share versus (\$1.6 million), or (\$0.04) in the prior year. A reconciliation of adjusted EBITDA and adjusted diluted EPS to the most directly comparable GAAP measures is provided at the end of this press release.

2023 Full Year Financial Results

	Years Ended	Dece	ember 31,	C	Change from I	Prior Year	% of 7	Total
	2023		2022		\$	%	2023	2022
Net Revenues							<u></u>	
Domestic Core	\$ 22,446,519	\$	21,205,204	\$	1,241,315	5.9%	78.7%	76.0%
International Core	4,596,097		4,164,714		431,383	10.4%	16.1%	14.9%
Novel Therapies	 1,475,050		2,526,119		(1,051,069)	(41.6%)	5.2%	9.1%
Total	\$ 28,517,666	\$	27,896,037	\$	621,629	2.2%		

Total net revenues increased \$0.6 million, or 2.2%, for the year ended December 31, 2023, as compared with the same period last year. Domestic Core growth of 5.9% was driven by volume growth in pumps and consumables attributed to share gains. International Core growth of 10.4% was driven by increased volume across several EU markets and the entry into multiple new geographic markets. The growth in Core offset a 41.6% year-over-year decline in Novel Therapies net revenues, primarily driven by the completion of a large milestone for a pharmaceutical NRE agreement in 2022.

Gross profit increased \$1.3 million or 8.7% for the year ended December 31, 2023, compared to the same period in 2022. Gross profit as a percentage of net revenues increased to 58.6% compared to 55.1% from the prior year. The increase was driven by improvements in manufacturing productivity and product mix. Allowing for a one-time inventory adjustment for a product discontinuation, adjusted gross margin for the year ended 2023 was 59.6%. A reconciliation of adjusted gross margin to the most directly comparable GAAP measure is provided at the end of this press release.

Selling, general and administrative expenses decreased \$0.2 million, or 1.2%, during the year ended December 31, 2023, compared to the same period last year, primarily driven by lower executive restructuring costs partially offset by an increase of new hires in commercial business development and medical affairs.

Research and development expenses increased \$0.8 million, or 15.9%, during the year ended December 31, 2023, compared with the same period last year, primarily due to compensation and expenses to support new hires focused on new product development.

Net loss for the full year 2023 was \$13.7 million, or \$(0.30) per diluted share, compared to a net loss of \$8.7 million, or \$(0.19) per diluted share for the same period of 2022. Net loss included a tax valuation allowance of \$6.0 million for the full year 2023. Adjusted EBITDA for the full year was (\$6.0) million, or (\$0.13) per diluted share versus (\$6.1) million or (\$0.14) in the prior year. A reconciliation of adjusted EBITDA and adjusted diluted EPS to the most directly comparable GAAP measures is provided at the end of this press release.

- 2 -

Cash and cash equivalents were \$11.5 million as of December 31, 2023, reflecting positive cash flow of \$0.7 million in the fourth quarter and a reduction of cash burn to \$5.9 million in 2023 from \$8 million in 2022.

Subsequent Events

The Company has entered into a \$5M revolving credit facility and a \$5M term loan facility with HSBC Bank USA, which replaced a \$3.5M revolving credit facility with a regional bank that expired in August of 2023. This new debt facility provides KORU Medical with additional capital flexibility to support a reserve for growth and strategic opportunities.

The Company has received an assessment report from its notified body in the EU, BSI, stating that, following BSI's review of technical documentation submitted by the Company in connection with a prior audit nonconformance, a recommendation for continued certification cannot be made. The Company has filed an appeal to this determination. Management believes that the Company's appeal will be successful in limiting the scope of the suspension to have minimal impact on the Company's revenues, if any.

Assumptions and Guidance for Full Year 2024

KORU Medical's guidance for the full year 2024 reflects numerous assumptions that could affect its business, based on the information management has as of this date. Management will discuss its outlook and several of its assumptions on its fourth quarter 2023 earnings call.

KORU Medical expects:

- Full year 2024 net revenue between \$31.2 and \$32.2 million, representing growth in the range of 10% to 13%
 - o SCIg market growth of mid-to- high single digits
 - o Three new Novel Therapies collaborations
 - Prefilled syringe market penetration of approximately 20%-25%
- Gross margin between 59% and 610% for the full year 2024
 - o Geographic expansion into lower ASP markets

- o Supply chain inflationary pricing pressure
- o Production line start-up in Q4 for new product introduction
- 2024 year-end cash balance greater than \$8.0 million
 - Operating expenses of approximately \$23.5-\$24.0 million, exclusive of stock compensation expense
 - Cash flow breakeven in in the fourth quarter of 2024 and cash flow positive for full year 2025
 - Ending cash balance is exclusive of the new \$10 million credit facility, which will be reserved for strategic growth capital opportunities

Conference Call and Webcast Details

The Company will host a live conference call and webcast to discuss these results and provide a corporate update on Wednesday, March 13, 2024, at 4:30 PM ET.

To participate in the call, please dial (844) 826-3035 (domestic) or (412) 317-5195 (international) and provide conference ID: 6447448. The live webcast will be available on the IR Calendar on the News/Events page of the Investors section of KORU Medical's website.

- 3 -

Non-GAAP Measures

This press release includes the non-GAAP financial measures "adjusted gross margin," "adjusted diluted EPS" and "adjusted EBITDA" that are not in accordance with, nor an alternate to, generally accepted accounting principles and may be different from non-GAAP measures used by other companies. These non-GAAP measures are not based on any comprehensive set of accounting rules or principles. Non-GAAP financial measures should not be considered a substitute for, or superior to, measures of financial performance prepared in accordance with GAAP. They are limited in value because they exclude charges that have a material effect on KORU Medical's reported results and, therefore, should not be relied upon as the sole financial measures to evaluate the Company's financial results. Non-GAAP financial measures are meant to supplement, and to be viewed in conjunction with, GAAP financial results. Reconciliations of the Company's non-GAAP measures are included at the end of this press release.

About KORU Medical Systems

KORU Medical develops, manufactures, and commercializes innovative and patient-centric large volume subcutaneous infusion solutions that improve quality of life for patients around the world. The FREEDOM Syringe Infusion System ("the FREEDOM System") currently includes the FREEDOM60® and FreedomEdge® Syringe Infusion Drivers, Precision Flow Rate TubingTM and HIgH-Flo Subcutaneous Safety Needle SetsTM. These devices are used for infusions administered in the home and alternate care settings. For more information, please visit www.korumedical.com.

Forward-looking Statements

This press release contains forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995 that involve risks and uncertainties. All statements that are not historical fact are forward-looking statements, including, but not limited to, financial guidance and expected operating performance for fiscal 2024. Forward-looking statements discuss the Company's current expectations and projections relating to its financial position, results of operations, plans, objectives, future performance, and business. Forward-looking statements can be identified by words such as "guidance", "expect", "plan", "believe" and "will". Actual results may differ materially from the results predicted and reported results should not be considered as an indication of future performance. The potential risks and uncertainties that could cause actual results to differ from the results predicted include, among others, uncertainties associated with SCIg market growth, prefilled syringe penetration, plasma supply, clinical trial activity and success, the Company's EU certification, the shift to increased healthcare delivery in the home, new patient diagnoses, customer ordering patterns, global health crises, innovation and competition, labor and supply price increases, inflationary impacts, labor supply, and those risks and uncertainties included under the captions "Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2023 which is on file with the SEC and available on our website at www.korumedical.com/investors and on the SEC website at www.sec.gov. All information provided in this release and in the attachments is as of March 13, 2024. Undue reliance should not be placed on the forward-looking statements in this press release, which are based on information available to us on the date hereof. We undertake no duty to update this information unless required by law.

Investor Contact:

Louisa Smith

investor@korumedical.com

- 4 -

KORU MEDICAL SYSTEMS, INC. BALANCE SHEETS

December 31, December 31, 2023 2022

CURRENT ACCETC				
CURRENT ASSETS Cash and cash equivalents	\$	11,482,240	•	17,408,257
1	Þ	11,482,240	Э	17,408,237
Accounts receivable less allowance for doubtful accounts of \$24,777 and \$21,459 for		4.045.211		2 550 001
December 31, 2023, and December 31, 2022, respectively Inventory		4,045,211 3,481,301		3,558,884 6,404,867
Other receivables				
		28,889		972,396
Prepaid expenses and other	_	1,218,288	_	1,457,232
TOTAL CURRENT ASSETS		20,255,929		29,801,636
Property and equipment, net		3,837,657		3,886,975
Intangible assets, net of accumulated amortization of \$390,341 and \$325,872 at December 31, 2023 and December 31, 2022, respectively		754,361		787,182
Operating lease right-of-use assets		3,514,055		3,786,545
Deferred income tax assets, net allowance for non-realization of deferred tax assets of		3,314,033		3,760,343
\$6,002,777 and zero for December 31, 2023 and December 31, 2022, respectively				3,967,480
Other assets		98,970		102,625
	\$	28,460,972	Φ	42,332,443
TOTAL ASSETS	<u> </u>	28,460,972	D	42,332,443
A A DAY PENDE A AND CITE CANNOT DEDUCATE A CONTROL OF THE CONTROL				
LIABILITIES AND STOCKHOLDERS' EQUITY				
CURRENT LIABILITIES				
Accounts payable	\$	975,193	\$	2,391,799
Accrued expenses		1,711,427		2,889,941
Note Payable		314,344		433,295
Other liabilities		512,520		257,337
Accrued payroll and related taxes		462,941		542,399
Finance lease liability – current		109,540		98,335
Operating lease liability – current		368,313		345,834
TOTAL CURRENT LIABILITIES		4,454,278		6,958,940
Finance lease liability, net current portion		316,623		394,283
Operating lease liability, net of current portion		3,336,300		3,653,257
TOTAL LIABILITIES		8,107,201		11,006,480
Commitments and contingencies (Refer to Note 8)				
STOCKHOLDERS' EQUITY				
Common stock, \$0.01 par value, 75,000,000 shares authorized, 49,089,864 and 48,861,891				
shares issued; 45,669,362 and 45,441,389 shares outstanding at December 31, 2023, and				
December 31, 2022, respectively		490,899		488,619
Additional paid-in capital		47,018,707		44,252,117
Treasury stock, 3,420,502 shares at December 31, 2023 and December 31, 2022, at cost		(3,843,562)		(3,843,562)
Retained Deficit	((23,312,273)		(9,571,211)
TOTAL STOCKHOLDERS' EQUITY		20,353,771		31,325,963
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY	\$	28,460,972	\$	42,332,443
C				

See accompanying Notes to Financial Statements.

- 5 -

KORU MEDICAL SYSTEMS, INC. STATEMENTS OF OPERATIONS

	Three Mon Decem				Twelve Mor		
	2023		2022	_	2023	_	2022
NET REVENUES	\$ 7,185,932	\$	7,344,681	\$	28,517,666	\$	27,896,037
Cost of goods sold	2,854,987		3,266,535		11,809,384		12,527,051
Gross Profit	4,330,946		4,078,146		16,708,282		15,368,986
OPERATING EXPENSES							
Selling, general and administrative	5,012,646		4,761,173		20,365,617		20,606,507
Research and development	1,287,515		1,640,732		5,742,254		4,956,215
Depreciation and amortization	228,340		187,658		870,390		587,137
Total Operating Expenses	6,528,501		6,589,563		26,978,261	_	26,149,859
Net Operating Loss	(2,197,555)		(2,511,417)		(10,269,979)		(10,780,873)
Non-Operating (Expense)/Income							
(Loss)/Gain on currency exchange	7,418		(977)		(5,124)		(39,874)
Loss on disposal of fixed assets, net	_		_		(59,807)		_
Interest (expense)/income, net	169,230		101,008		561,328		145,587
TOTAL OTHER (EXPENSE)/INCOME	176,648		100,031		496,397		105,713
LOSS BEFORE INCOME TAXES	(2,020,907)		(2,411,386)		(9,773,582)		(10,675,160)
Income Tax Benefit/(Expense)	(5,445,123)	_	434,659	_	(3,967,480)	_	2,014,018
NET LOSS	\$ (7,466,029)	\$	(1,976,727)	\$	(13,741,062)	\$	(8,661,142)

NET LOSS PER SHARE				
Basic	\$ (0.16)	\$ (0.04)	\$ (0.30)	
Diluted	\$ (0.16)	\$ (0.04)	\$ (0.30)	\$ (0.19)
WEIGHTED AVERAGE NUMBER OF COMMON SHARES OUTSTANDING				
Basic	45,669,691	45,372,132	45,601,346	45,002,074
Diluted	45,669,691	45,372,132	45,601,346	45,002,074

See accompanying Notes to Financial Statements.

- 6 -

KORU MEDICAL SYSTEMS, INC. STATEMENTS OF CASH FLOWS

	For the Yes	
	2023	2022
CASH FLOWS FROM OPERATING ACTIVITIES	0 (12 741 062)	A (0.661.140)
Net Loss	\$ (13,741,062)	\$ (8,661,142)
Adjustments to reconcile net loss to net cash used in operating activities:	2.760.070	2.070.427
Stock-based compensation expense Depreciation and amortization	2,768,870	3,079,427
Loss on disposal of fixed assets	870,390 59,807	587,137
Deferred income taxes	(2,035,297)	(2.026.226)
Allowance for non-realization of deferred tax asset	6,002,777	(2,026,226)
ROU landlord credit	(21,988)	212,546
Changes in operating assets and liabilities:	(21,988)	212,340
(Increase)/Decrease in accounts receivable	(486,327)	34,002
Decrease/(Increase) in inventory	2,923,566	(298,529)
Decrease/(Increase) in other receivables	943,507	(254,176)
Decrease in prepaid expenses and other assets	242,599	28,776
(Decrease)/Increase in accounts payable	(1,416,606)	1,164,266
(Decrease)/Increase in accrued payroll and related taxes	(79,458)	381,796
Increase in other liabilities	255,183	167,337
(Decrease)/Increase in accrued expenses	(1,178,514)	180,237
NET CASH USED IN OPERATING ACTIVITIES	(4,892,553)	(5,404,549)
CASH FLOWS FROM INVESTING ACTIVITIES		
Purchases of property and equipment	(782,949)	(2,761,056)
Purchases of intangible assets	(31,648)	(40,512)
NET CASH USED IN INVESTING ACTIVITIES	(814,597)	(2,801,568)
CASH FLOWS FROM FINANCING ACTIVITIES		
Proceeds from issuance of equity	_	406,623
Borrowings from indebtedness	565,172	783,799
Payments on indebtedness	(684,123)	(859,087)
Finance lease ROU asset	(33,461)	_
Payments on finance lease liability	(66,455)	(51,850)
NET CASH (USED IN) PROVIDED BY FINANCING ACTIVITIES	(218,867)	279,485
NET DECRE 4 CE NA CA CIVANDE CA CIVANDE CA CIVANDA DA CIVADA	(5.00 < 0.15)	(7.026.622)
NET DECREASE IN CASH AND CASH EQUIVALENTS	(5,926,017)	(7,926,632)
CASH AND CASH EQUIVALENTS, BEGINNING OF YEAR	17,408,257	25,334,889
CASH AND CASH EQUIVALENTS, END OF YEAR	\$ 11,482,240	\$ 17,408,257
Supplemental Information		
Cash paid during the years for:		
Interest	\$ 50,832	\$ 28,490
Income taxes	\$ 3,160	\$ —
Schedule of Non-Cash Operating, Investing and Financing Activities:		
Issuance of common stock as compensation	\$ 446,349	\$ 513,082
-		

See accompanying Notes to Financial Statements.

Reconciliation of GAAP Net (Loss)	Three Mon Decemb	 	Twelve Months Ended December 31,			
to Non-GAAP Adjusted EBITDA:	 2023	2022	2023		2022	
GAAP Net Loss	\$ (7,466,029)	\$ (1,976,727)	\$ (13,741,062)	\$	(8,661,142)	
Tax (Benefit)/Expense	(557,654)	(434,659)	(2,035,297)		(2,014,018)	
Valuation Allowance for DTA	6,002,777	_	6,002,777		_	
Depreciation and Amortization	228,340	187,658	870,390		587,137	
Interest (Income), Net	(169,230)	(101,009)	(561,328)		(145,587)	
Reorganization Charges	329,869	_	329,869		765,433	
Manufacturing Initiative Expenses	_	184,343	55,361		293,229	
Product Discontinuance	280,000	_	280,000		_	
Stock-based Compensation Expense	389,256	588,654	2,768,869		3,079,426	
Adjusted EBITDA	\$ (962,671)	\$ (1,551,740)	\$ (6,030,421)	\$	(6,095,522)	

Reconciliation of Reported Diluted EPS	Three Months Er December 31,		Twelve Months Ended December 31,			
to Non-GAAP Adjusted Diluted EPS:	 2023	2022	2023	2022		
Reported Diluted Earnings Per Share	\$ (0.16) \$	(0.04) \$	(0.30) \$	(0.19)		
Tax (Benefit)/Expense	(0.01)	(0.01)	(0.04)	(0.05)		
Valuation Allowance for DTA	0.13	_	0.13	_		
Depreciation and Amortization	_		0.02	0.01		
Interest (Income)/Expense, Net	_	_	(0.01)	_		
Reorganization Charges	0.01	_	0.01	0.02		
Manufacturing Initiative Expenses	_	_	_	_		
Product Discontinuance	_	_	_	_		
Stock-based Compensation Expense	 0.01	0.01	0.06	0.07		
Adjusted Diluted Earnings Per Share	\$ (0.02) \$	(0.04) \$	(0.13) \$	(0.14)		

Reconciliation of GAAP Gross	Three Mon Decem		Nine Months Ended December 31,			
Profit to Adjusted Gross Margin	2023	2022	2023	2022		
Reported Gross Profit stated as a percentage of Net						
Revenues (Gross Margin)	60.3%	55.6%	58.6%	55.1%		
Product Discontinuance	2.8%	0.0%	1.0%	0.0%		
Adjusted Gross Profit stated as a percentage of Net						
Revenues (Adjusted Gross Margin)	63.1%	55.6%	59.6%	55.1%		

^{*}Numbers presented are rounded to the nearest whole cent and percentage

Reorganization Charges. We have excluded the effect of reorganization charges in calculating our non-GAAP measures. In 2022 and 2023 we incurred severance expenses related to the reorganization of the leadership team, which we would not have otherwise incurred in periods presented as part of continuing operations.

Manufacturing Initiative Expenses. We have excluded the effect of expenses related to creating manufacturing efficiencies in calculating our non-GAAP measures. We incurred expenses in connection with these initiatives which we would not have otherwise incurred in periods presented as part of our continuing operations.

Stock-based Compensation Expense. We have excluded the effect of stock-based compensation expense in calculating our non-GAAP measures. We record non-cash compensation expenses related to grants of options and restricted shares for executives, employees and consultants, and grants of shares to our board of directors. Depending upon the size, timing and the terms of the grants, the non-cash compensation expense may vary significantly but will recur in future periods.

Allowance for nonrealization of deferred tax assets (DTA). We have excluded the effect of recording a full valuation allowance on our deferred tax assets in the fourth quarter ended 2023 in the amount of \$6.0 million to illustrate the impact on net loss and earnings per share.

Product Discontinuation. We have excluded the effect of disposing of a non-launched inventoried product that has no value in our core business product portfolio.